CONFÉRENCE DES NATIONS UNIES SUR LE COMMERCE ET LE DÉVELOPPEMENT



UNITED NATIONS CONFERENCE ON TRADE AND DEVELOPMENT

# IMPACT OF GLOBAL SLOWDOWN ON INDIA'S EXPORTS AND EMPLOYMENT

**By UNCTAD India Team<sup>1</sup>** 

May 2009

## **Draft for Comments**

Prepared under UNCTAD- Govt. of India- DFID Project 'Strategies and Preparedness for Trade and Globalization in India'. The views expressed in this paper do not reflect the views of UNCTAD or it's member states.

<sup>&</sup>lt;sup>1</sup> UNCTAD-India team-Abhijit Das (Deputy Project Coordinator), Rashmi Banga (Senior Economist), Shahid Ahmed (Senior Economist) Ramaa Sambamurthy (Consultant) and Dinesh Kumar (Consultant). Research assistance provided by Ratika Bhanot is highly appreciated. We are grateful to Mrs. Lakshmi Puri, Director DITC, UNCTAD, and Mr. Bonapas Onguglo, UNCTAD for comments and suggestions.

## TABLE OF CONTENT

Executive Summary	7
1. Introduction	. 12
2. Trends in India's Total Exports	. 17
2.1 Trends in India's Merchandise Exports	. 17
2.1.1 Growth of India's Merchandise Exports	. 17
2.1.2 Composition of India's Export Basket	. 19
2.1.3 Direction of India's Exports	. 20
2.2 Trends in India's Services Exports	. 22
2.2.1 Growth in India's Services Exports Overtime	. 22
2.2.2. Composition of India's Services Exports	. 23
2.2.3 Direction of India's Exports of Services	. 26
2.3. Trends in India's Imports of Goods and Services	. 28
3. Impact of Slowdown on India's Exports	. 31
3.1 Methodology and Data	. 32
3.2 India's Income Elasticity of Total Exports	. 33
3.3 India's Income Elasticity of Sectoral Exports	. 34
4. Forecast of India's Exports using Income Elasticity of Exports	. 37
5. Impact of Slowdown on Employment through International Trade	. 39
6. Identification of Sectors for Employment Generation	. 41
7. Conclusions and Mitigating Strategies	. 42
ANNEX I: Results of Stationarity Tests	<b>. 49</b>
ANNEX II: New and Potential Products for Exports in Developing Countries	. 52

## LIST OF TABLES

Table 1 Growth in India's Trade (In Real Terms): 2005-06-2007-08 (%)	14
Table 2: Growth in India's Sectoral Exports in 2008 over 2004 in Major Ten Sectors	s 18
Table 3: Change in Composition of India's Export Basket: 2004-2008	19
Table 4: Share of Region/Country in India's Exports: 1990-91 to 2007-08	21
Table 5: India's Total Exports of Services to World and U.S.	23
Table 6: Composition of India's Exports of Services	23
Table 7: Services Exports of US and Share in Global Indian Services Exports	26
Table 8: India's Oil imports and Rates of growth (%)	29
Table 9: India's Non-Oil imports and Rates of growth (%).	29
Table 10: Income Elasticities of India's Export Demand	34
Table 11: Price and Income Elasticities for India's Major Sectors of Exports	35
Table 12: Projected Real GDP Growth (%): 2009 and 2010	37
Table 13: Forecasted Total Merchandise Export Growth Sectoral Export Growths: 2	.008-
09 and 2009-2010	37
Table 14: Impact of Slowdown on Employment: 2008-09 to 2010-11	40
Table 15: Employment Multipliers based on input-Output Matrix of 2004-04	41
Table 16: Potential Gain for India from Export of New and Potential Products	43

## LIST OF FIGURES

Figure 1: Slowdown in GDP Growth Rates: 2006-2008	13
Figure 2: India's Monthly Exports and Imports in 2008-09	15
Figure 3: Decline in Growth Rates of India's Merchandise Exports to World and to	U.S.:
2005-2008	17
Figure 4: India's Export Growth: 2005-06 to 2008-09	18
Figure 5: Change in Composition of India's Export Basket: 2004-2008	20
Figure 6: Growth of India's Exports of Services to World and U.S.: 2002-2008	22
Figure 7: India's Exports, Imports and Trade Balance: 2000-01 to 2007-08	28
Figure 8: Composition of India's Import Basket: 2004-05 to 2008-09	29

# IMPACT OF GLOBAL SLOWDOWN ON INDIA'S EXPORTS AND EMPLOYMENT

## **Executive Summary**

Riding on the back of brisk growth in the global economy since 2002, India's exports had witnessed a phenomenal three-fold increase during the period 2002-03 to 2007-08. But this powerful dynamo for employment generation is now threatened by rapid contraction in global demand and weakening labour markets. It is a major challenge for India to implement strategies which not only mitigate the adverse impact of global slowdown on its exports but also build the resilience of the economy to such future shocks. However, for designing such strategies there is a need to assess the extent to which global slowdown may impact India's total exports, and more importantly, identify the sectors which are likely to be more adversely affected by the slowdown.

In this context, the study forecasts the impact of slowdown in global GDP on India's total exports and exports of ten major sectors and estimates economy wide and sectoral employment impact in 2009-10 and 2010-11. It also identifies vulnerable sectors with high potential for employment generation for immediate policy interventions. Further, the study undertakes a detailed competitiveness analysis at six digit levels to identify new and potential exports in countries/ regions such as West Asia, ASEAN, Australia and Brazil, which are expected to recover faster than other economies. Some short term measures have also been suggested for cushioning the adverse impact of global slowdown on exporters.

Global demand plays an important role in determining export growth of a product. The impact of slowdown in global demand on a country's exports will largely be determined by income elasticity of demand of the product. Accordingly, the study estimates income elasticity for India's total exports and its sectoral components. These income elasticities, in conjunction with GDP growth forecasts for 2009 and 2010 (provided by OECD Economic Outlook, March 2009) are used to estimate India's total and sectoral export growth.

The results show that India's exports to world are very responsive to income changes. A 1% decline in GDP growth of world will lead to 1.88% decline in India's growth of exports to world. Estimates of income elasticities of ten major export sectors of India (which are around 95% of total India's exports) show that it is high for sectors such as petroleum products, ores and minerals, gems and jewellery, chemical products and engineering products. India's traditional export sectors like textiles leather and plantation have relatively low income elasticity, with lowest being for plantation.

Along with income elasiticity, price competitiveness may also determine the impact of slowdown on exports. If the products exported are less price sensitive, during slowdown the option of lowering prices to maintain existing market shares may not be feasible. Sectors which have high income elasticity but low price elasticity are therefore relatively more vulnerable sectors of the economy in terms of impact of global slowdown. Two

such sectors identified are gems and jewellery and textiles. These require targeted interventions.

#### Forecast of India's Exports using Income Elasticity of Exports

Using the income elasticities for export demand for India and the forecasted change in global GDP growth, India's total export growth and sectoral export growth for ten major sectors has been estimated for the year 2009-10 and from 2010-11 (till December 2010). The forecasted slowdown of GDP growth as provided by OECD, Economic Outlook (March 2009) are used.

<u>The results show that India's total exports will grow by -2.2% in 2009-10</u>, which implies that there will be almost flat growth, marginally tending towards a negative growth. Most of the sectors experience a negative growth rate. Positive growth in exports is forecasted for plantation, agriculture sector and engineering & electronics sector. It should be noted that though positive growth rates of exports in agricultural products has been forecasted, it is much lower than the 55% export growth in 2007-08. The forecasts also show that petroleum products will experience the maximum decline in export growth followed by gems and jewellery, ores and minerals and textiles and textile products.

Export growth is likely to experience a significant recovery and <u>increase to 8.3% in 2010-11</u>. All sectors are projected to experience a positive export growth in this period with sectors such as agricultural products, plantation, engineering, chemicals and petroleum products reaching their initial level of exports of 2007-08.

It should be noted that these estimates are critically based on the predictions for global GDP growth in 2009 and 2010.

#### Impact of Global Slowdown on Employment

The predicted overall export growth for the years 2009-10 and 2010-11 and sectoral export growths have been used to estimate the impact of global slowdown on employment in the economy. The estimates show that in the year 2008-09, due to negative export growth in sectors such as textiles, gems and jewellery, ores and minerals, etc, the total job loss in India was of around 1.16 million. However, the net employment created by exports in this year was positive, i.e., 1.25 million as many sectors experienced positive export growth. The net employment is sum total of jobs created and lost in different sectors overtime. In the year 2009-10, export growth is predicted to be -2.2%, and the total job loss is estimated to be around 1.3 million. But, since export growth is positive for some sectors like plantation and these sectors have high employment multipliers, the net employment loss is estimated to be 0.7 million.

For the year 2010-11, estimation could be done only for three quarters, i.e., till December 2010 as GDP growth predictions are not available beyond that period. Using the predicted export growth of 8.3%, the total employment generated in the economy is estimated to be 5.22 million, indicating that the loss in employment due to decline in exports in 2009-10 will be compensated in 2010-11.

#### Mitigating Strategies

To build resilience of the economy to trade shocks and improve competitiveness of the exports, it would be useful for the government to consider mitigating strategies. This study suggests five specific mitigating strategies relating to (a) diversification of exports to new geographical destinations and new products; (b) simplification in customs procedures for reducing transaction costs; (c) examination of the likely impact of antidumping and safeguard duties imposed by India on down-stream user industries; (d) measures aimed at assisting exporters to retain their market presence during the crisis period; and (e) expeditious multilateral examination of adverse impact of bailouts and stimulus packages and prompt remedies.

In markets which are expected to recover fast (ASEAN, Australia, Brazil, Korea, South Africa and West Asia) competitiveness analysis at a disaggregated level has been undertaken of India with respect to the importing country and its five major trading partners. This analysis identifies products in which India has the potential to significantly increase its exports from the current level (potential products) or start export of new product. Around 958 products have been identified. It is found that India has the potential to increase its exports of new and potential products by almost 21%, i.e., by USD 35 billion.

With profit margins shrinking globally, cost competitiveness would be an important determinant for retaining or acquiring a share in export markets. In an attempt to reduce some of the transaction costs associated with international trade, the government has been simplifying its customs procedures over the past few years. While this is a continuing process, it needs to gather significant additional momentum.

India has been a major user of anti-dumping measures over the past few years, there has been a significant increase in the number of new anti-dumping and safeguard investigations initiated from October 2008 onwards. In the context of current global slowdown, it may be beneficial for the economy as a whole if a detailed economic analysis on the likely impact of the duties on downstream user industry is undertaken, prior to imposition of the duties.

With economic recovery being predicted for 2010, it is important that India's exporters do not withdraw from the export market in the intervening period of down turn, if they are to take advantage of export opportunities during the period of recovery. Government could consider a two-pronged approach for supporting exporters to retain their presence in foreign markets. It could support exporters through incentives such as easing trade financing. However, as export related incentives can be neutralized or offset by the importing country through imposition of countervailing duty, an attempt could be made at the multilateral level to explore the possibility of a stand-still on countervailing duties that might otherwise arise from incentives given by developing countries.

A large number of stimulus and bail-out packages have been offered across the world. The government could consider putting into place a mechanism, at least in the short term, for constantly reviewing the implementation of these packages and identifying measures, if any, which may have an adverse impact on India's export interest. In addition to implementing the mitigation strategies outlined above, there is a need to develop and implement long term measures that would ensure sustained export growth which are not impeded by adverse developments in big foreign markets.

## IMPACT OF GLOBAL SLOWDOWN ON INDIA'S EXPORTS AND EMPLOYMENT

#### 1. Introduction

Due to increased integration of the world markets, transmission of economic crisis from one country to the rest of the world has become smoother. The larger the country, where the crisis originates, the greater is the impact on other countries. US, one of the largest economy in the world, both in terms of its share in world GDP (27%) and global imports (17%) experienced the sub-prime mortgage collapse in August 2007. This was followed by the reversal of the housing boom in other industrialized economies, which had a ripple effect all around the world. Furthermore, integrated financial sectors unmasked other weaknesses in the global financial system as a result of which some of the financial products and instruments became so complex and twisted, that as things started to unravel, trust in the whole system started to fail. Stock markets crashed all over the world, with declines ranging from 35-40% over the past 12 to 18 months in developed countries and even more in most emerging markets.

Crisis which emerged in the financial market creeped into the real sector of countries around the world through different channels. Credit squeezes due to instable financial instruments and stock market bursts led to contraction of output growth in the advanced financially integrated countries and resulted in lower real demand for capital and consumer goods in the advanced countries. Further, lower capital flows and investments into developing countries; lower remittances and savings; and lower commodity prices coupled with weak dollar aggravated the recession.

One of the most important channel through which the financial crisis erupting in US and in other advanced countries has been transmitted to developing countries is international trade. Apart from the direct impact of lower demand for exports of developing countries in bilateral trade with advanced economies, impact of slowdown can be transmitted through three other major channels of trade. Firstly, through third market effects, i.e., "echo effects", as referred to in the literature, which works through the trading partners of the country where slowdown occurs. Apart from the direct effects of lowering of exports to the country experiencing slowdown, there is an indirect effect through lower demand from trading partners of this country as their GDP growth also slows down due to lowering of the demand for their exports. This leads to a second round of slowdown of demand for exports of developing countries. Secondly, the impact of slowdown may be transmitted through the 'supply chain effect'. The international vertical supply chains are adversely affected and developing countries which are a part of these supply chains may feel the impact of lowering of demand for their exports to other developing countries which in turns leads to lower exports. *Thirdly*, in addition to these, trade finance squeeze due to tighter financial markets can lead to substantial supply side effects.

However, the impact of slowdown may be felt differently by different countries depending on the nature of their exportable products, destination country of exports and

the overall dependence of the economy on exports. Further, higher the income elasticity of demand for a country's exports, higher will be the adverse impact of lower GDP growth of its trading partners.

One of the unique features of US economy is its high income elasticity of imports<sup>2</sup>. Three decades of econometric modeling<sup>3</sup> show that income elasticity of imports in US is greater than 1. While estimates vary, it is generally found that for every 1% increase in US income, import demand increases by 2.2%. The implication of this is clear: a 1% slowdown of GDP in US will decrease the import demand by 2.2%. This can transmit the slowdown of US rapidly into the countries which have US as a major market for their exports.

India is one of the many developing countries which have relied heavily on US and other advanced countries for its exports. In 2007, around 17% of India's exports sought US markets, while 29% were directed to G7 countries<sup>4</sup> and around 58% of the exports were directed towards advanced countries (as defined by IMF). Given such heavy reliance on advanced countries' markets the impact of slowdown in these countries is being felt heavily in India's trade sector.

While GDP growth of the world has declined from 4.3% in 2006 to 2.2% in 2008, it has declined much faster in advanced countries like U.S (from 2.8% to 1.1.%) and European Union (3.0% to 0.7%) (Figure 1).





Along with global slow down in the growth of GDP, there has also been a substantial decline in world trade which may result in echo effects. The world's real trade growth (corrected for prices) declined from 9.5% in 2006 to 6.9% in 2007 and further to 2.5% in

Source: OECD and IMF

<sup>&</sup>lt;sup>2</sup> where income elasticity of import/export is defined as percentage change in growth of imports/exports for one percentage change in growth in incomes or GDP.

<sup>&</sup>lt;sup>3</sup> Magee (1975), Sawyer and Sprinkle (1996), Marquez (2001)

<sup>&</sup>lt;sup>4</sup> G7 countries are as defined by IMF.

2008. Amidst the global economic slowdown, O.E.C.D. projects global economy to grow by -2.7% in 2009 and 1.2% in 2010. The U.S. economy is expected to experience a negative growth, i.e., -4% in 2009 and 0% growth in 2010. EU area is expected to also experience a negative growth of -4.1% in 2009 with a continued negative growth of -0.3% in 2010. China is expected to grow at 6.3% and growth forecast for the Indian economy is at 4.30% for the year 2009.

Snapshot of the World Economy													
Real GDP (% change)	2006	2007	2008	2009P	2010P								
United States	2.80	2.00	1.10	-4.00	0.00								
Euro Area	3.00	2.60	0.70	-4.10	-0.30								
Canada	3.10	2.70	0.50	-3.00	0.30								
United Kingdom	2.80	3.00	0.70	-3.70	-0.20								
Japan	2.00	2.40	-0.60	-6.60	-0.50								
India	9.70	9.00	6.00	4.30	5.80								
China	11.60	13.00	9.00	6.30	8.50								

Source: O.E.C.D. Database

Although India is expected to grow, India has not been able to remain insulated in this global decline, especially in the trade sector. A close look at India's trade sector indicates that in real terms growth in India's exports and imports in both goods and services has declined (Table 1). Growth in exports of goods in real term declined from 17.8% in 2006-07 to 5.4% in 2007-08. Maximum decline is witnessed in growth of exports of services which grew at the rate of 26.8% in 2005-06, but experienced a negative growth of -1.8% in 2007-08. Growth in imports of goods, declined from 25.2% in 2005-906 to 10.6% in 2007-08. Surprisingly, growth in private remittances in real terms has shown a marked improvement 10% in 2006-07 to 24.1% in 2007-08.

GDP growth of India was estimated to be 9.2% in 2005-06, which increased to 9.7% in 2006-07 but declined to 9.2% in 2007-08 and is expected to decline further to 7.2% in 2008-09 according to advance estimates.

	2005-06	2006-07	2007-08
Exports of Goods	17.2	17.8	5.4
Exports of Services	26.8	27.4	-1.8
Imports of Goods	25.2	17.9	10.6
Imports of Services	17.8	24.0	-3.7
Private Remittances	12.9	10.0	24.1
Real GDP at Market			
Prices	9.2	9.7	9.2

Table 1 Growth in India's Trade (In Real Terms): 2005-06-2007-08 (%)

Source: National Accounts Statistics, CSO and RBI

The slowdown in the trade sector post April 2008 is even more explicit (Figure 2). Exports have declined continuously since July 2008 except in the month of December.. They declined from USD 17,095 millions in July 2008 to USD 11,516 millions in March 2009, which accounts for almost 33% decline. While imports declined from USD 29,211

millions in August 2008 to USD 15,561 millions in March 2009 which accounts for almost 47% decline. But in terms of balance of trade (BOT), the deficit reduced from USD 8,747 millions in April to USD 4,045 millions in March.



Figure 2: India's Monthly Exports and Imports in 2008-09

Given the high dependence of Indian economy on its external trade sector, where exports of goods and services (less export related imports) is around 20% of GDP, a slowdown in trade sector can have adverse ripple effects in the economy. More importantly, it can lead to job losses and increase the number of poor in the country. The job losses may be direct due to contraction in output in the exportable sectors and indirect, which may occur due to decline in output of the sectors which provide inputs to the exportable sectors. The increase in cheaper imports, particularly of inferior goods (whose demand increases with lowering of incomes), can further add to contraction of output and employment in the economy.

In order to diminish the adverse impacts of global slowdown on Indian economy and improve the economy's resilience to external shocks to its trade sector, overall and sector-specific strategies need to be designed. However, for designing such strategies there is a need to assess the extent to which global slowdown may impact India's total exports and more importantly, identify the sectors which are likely to be more adversely affected by the slowdown. For this purpose, the study attempts to forecast the impact of slowdown in global GDP on India's total exports and sectoral exports to the world have been estimated using econometric models. Using these income demand elasticities, the impact of lower growth in global GDP (which have been forecasted by OECD) on India's exports in the period 2009-2010 has been estimated. The changes in sectoral exports have then been used to estimate the direct as well indirect impact on sectoral employment in the economy in the year 2009-2010.

An important contribution of the study is a detailed analysis of India's competitiveness at six-digit HS codes in markets of developing countries. The projected slowdown in developing countries is much lower than those in the advanced economies. Therefore a significant step in terms of mitigating the adverse impact of slowdown on India's exports will be to diversify Indian export basket and markets. The study identifies new and potential exports<sup>5</sup> in countries/ regions such as China, West Asia, ASEAN, Australia and Brazil. It estimates the likely share that India may get in case it is able to diversify into new products and new markets. Further, the study makes suggestions for mitigating the adverse impact of global slowdown on the Indian economy..

The study is organised as follows: Section 2 discusses trends in India's total and sectoral exports and imports using trade data. Section 3 presents the results with respect to global income demand elasticities for India's total and sectoral exports and estimates the impact of lower growth of global GDP on India's sectoral exports in 2009-2010 and 2010-11. Section 4, presents the estimates of impact of predicted export growth on total and sectoral employment for the years 2008-09, 2009-2010 and 2010-11. Section 5 discusses the mitigating strategies and identifies new products and new markets for India's exports. Section 6 identifies sectors for employment generation; section 7 concludes.

<sup>&</sup>lt;sup>5</sup> New exports refers to products where India's has competitive advantage in a market but is currently not exporting while potential exports refers to products where India is exporting but has the potential to increase its exports.

## 2. Trends in India's Total Exports

The extent to which global slowdown may impact a country's exports depends largely on the number of trading partners of the country and the composition of its export basket. High dependence on few markets and few exportable products may increase the severity of the impact of slowdown on exports, both in terms of coverage and depth. In order to assess the extent of the impact of global slowdown on India's exports, we examine the trends overtime in composition of India's export basket and its direction.

## 2.1 Trends in India's Merchandise Exports

#### 2.1.1 Growth of India's Merchandise Exports

India's global merchandise exports were growing at an impressive rate before the financial crisis occurred in the U.S. The global merchandise exports increased from USD 79 billion in 2004 to USD 145 billion in 2007, representing an average annual growth rate of 20%. However, slowdown of U.S. economy led not only to a decline in India's bilateral merchandise exports to U.S. but also its exports to the world. However, the decline in growth rate of merchandise exports to U.S was much higher than the decline in growth rate of total merchandise exports to the world (Figure 3). India's global exports which grew at 29.5% in 2005, over the preceding year, grew at a lower rate, i.e., 23.6% in 2008, while the decline in growth rate of exports to U.S. was much higher, i.e., from 32.3% in 2005 to 6.15% in 2008.

# Figure 3: Decline in Growth Rates of India's Merchandise Exports to World and to U.S.: 2005-2008



From the above trend, two facts emerge. First, India's export growth to the world has been more buoyant than its export growth to US and second, the decline in growth of exports to U.S. began in 2006, i.e., before the slowdown. Thus, the lack of buoyancy of

India's export growth to U.S may have cushioned and delayed some of the adverse effect of U.S slowdown on India's exports.

The quarterly trend shows that the export growth became negative for the first time since 2005-06 in the third quarter (Oct-Dec) of 2008-09 (-13.5%). Further, in the last quarter of 2008-09 (Jan-March 2009) there was a much steeper fall of -27.7%. <u>The impact of slowdown has therefore been felt in India from October 2008 onwards.</u>



#### Figure 4: India's Export Growth: 2005-06 to 2008-09

Comparing export growth in past five years across sectors, we find that exports have grown maximum in case of petroleum products, which experienced an export growth of 341% in 2008 as compared to 2004 (Table 2). Exports of engineering goods, agricultural products and chemicals and related products have also grown significantly, i.e., more than 100% in this period. However, traditional export sectors of India, like textiles and products, Leather and products and gems and jewellery have witnessed a comparatively lower export growth. India's total exports increased by 110% in 2008 over 2004.

S.No		(%)
1	PETROLEUM PRODUCTS	341.8
2	ENGINEERING GOODS	165.3
3	AGRI & ALLIED PRDTS	146.2
4	CHEMICALS & RELATED PRODUCTS	109.2
5	ORES & MINERALS	95.9
6	TEXTILES	52.9
7	LEATHER & MNFRS	34.4
8	PLANTATION	33.5
9	GEMS & JEWELLERY	32.3
10	MARINE PRODUCTS	7.4
	Total	110.1

Table 2: Growth in India's Sectoral Exports in 2008 over 2004 in Major Ten Sectors

#### 2.1.2 Composition of India's Export Basket

Concentration on few exportable products may worsen the impact of global slowdown on the exports of a country, especially if these products are those whose demand is closely related to incomes of the people, in other words, if these are not necessity products. India's traditional exports have constituted of items such as textile products, gems and jewellery, tea & coffee and leather & leather products. It is important to trace the extent of diversification of the export basket overtime. The trends show that there has been some diversification in composition of India's export basket overtime. However, there still remains large scope for further diversification.

As seen in Table 3 and Figure 5, the share of petroleum products (including rubber and plastic products) in India's export basket has been increasing since 2004. India exported USD 6.8 billion worth petroleum products in 2004 which increased to USD 23.6 billion in 2007 and further to 30.4 billion in 2008 and its share increased from 8.6% to 18.1%. Interestingly, the share of textiles, which has been predominant sector in export basket in 2004 (16.8%), has been declining continously and it reached to 12% in 2008. Engineering goods, representing a very broad category, continues to be a sector with highest share in India's export basket. Its share has further increased from 19.7% in 2004 to 25% in 2008. Share of chemical and chemical products has remained same overtime (13.7%) while share of gems and jewellery has declined from 18% in 2004 to around 11% in 2008.

Interestingly, exports of India's agricultural products has been rising steadily from USD 6.0 billion in 2004 to USD 14.9 billion, though their share in India's export basket still remains low (around 9%). Although exports of ores and minerals have nearly doubled from USD 4.3 billion to USD 8.4 billion in 2008, share of this sector in export basket remains around 5%. Marine and plantations have a share of around 1% share, which has not changed overtime. Plantation has less than 1% share in India's export basket.

S.No		2004	2006	2008
1	ENGINEERING GOODS	19.70	21.79	24.87
2	PETROLEUM PRODUCTS	8.63	14.96	18.15
3	CHEMICALS & RELATED PRODUCTS	13.72	13.67	13.65
4	TEXTILES	16.77	15.40	12.20
5	GEMS & JEWELLERY	17.84	12.72	11.23
6	AGRI & ALLIED PRDTS	7.63	6.78	8.94
7	ORES & MINERALS	5.42	4.78	5.05
8	LEATHER & MNFRS	3.20	2.66	2.05
9	MARINE PRODUCTS	1.71	1.40	0.87
10	PLANTATION	0.99	0.94	0.63
	Total	100.00	100.00	100.00

Table 3: Change in Composition of India's Export Basket: 2004-2008



Figure 5: Change in Composition of India's Export Basket: 2004-2008

The above trends in composition of India's export basket show that India's export basket has diversified in past five years with engineering goods, petroleum products and chemical products increasing their share in the export basket, while traditional exports like textiles, gems and jewellery and leather and leather products losing their shares.

There exists considerable scope for further diversification of India's export basket in terms of its composition. A disaggregated level analysis at HS 6-digit level on the number of products accounting for 50% of the total trade brings out this point more clearly. A quinqennial comparison over the period starting from 1996-97 to 2007-08 shows that in 2007-08 around 34 products at 6-digit level accounted for 50% of global exports of India. This number has declined from around 45 products in 2000-01. There exists large scope to further diversify India's export basket by identifying at products six-digit level where India may have higher competitiveness in production.

#### 2.1.3 Direction of India's Exports

As discussed earlier, the extent to which global slowdown affects a country's exports is likely to be determined by the extent of dependence of exports on trading partners affected by the slowdown. Concentration of exports in few markets which are facing slowdown may hasten the transmission of adverse impact of slowdown.

In the 1990s, more than half of India's exports were directed towards OECD markets, with 28% directed to EU markets and around 15% to U.S. Around 16% went to Russia and similar percentage to developing countries with Asian markets being more dominant (Table 4). However, over time there has been some diversification in terms of direction of India's exports. Share of EU has declined from 28% in 1995-96 to 20% in 2007-08, while share of US has declined from 17.4% in 1995-96 to 13% in 2007-08. Share of U.A.E has increased from 4.5% in 1995-96 to 9.7% in 2007-08. There has been considerable increase in share of Asian developing countries in India's export basket, from 23% in 1995-06 to 31.5% in 2007-08. Share of Africa has also increased over time. It is

interesting to note that <u>share of developing countries in India's exports increased from</u> 17% in 1990-91 to 42% in 2007-08.

The fact that India was able to diversify its exports to different countries has helped in softening the impact of global slowdown on India's exports. <u>However, bulk of India's exports, i.e., 33% is still directed towards EU and U.S. There is a need to further diversify exports in terms of new destinations.</u>

Group /					1990-	1995-	2000-	2005-	2007-
Country					91	96	01	06	08
Ι.	OECD				56.5	55.7	52.7	44.5	38.8
	Α.	EU			27.5	27.4	23.4	21.7	20.2
		North							
	В.	America						17.8	13.8
			1	Canada	0.9	1.0	1.5	1.0	0.8
			2	U.S.A	14.7	17.4	20.9	16.8	13.0
		Asia and							
	C.	Oceania					5.1	3.3	3.1
		of which:							
			1	Australia	1.0	1.2	0.9	0.8	0.7
			2	Japan	9.3	7.0	4.0	2.4	2.2
	_	Other OECD	)						
	D.	countries					1.9	1.6	1.7
11.	OPEC				5.6	9.7	10.9	14.8	16.5
	of which:								
	1	U.A.E.			2.4	4.5	5.8	8.3	9.7
	Eastern								
III.	Europe				17.9	4.2	3.0	1.9	2.1
	of which:								
	1.	Russia			16.1	3.3	2.0	0.7	0.6
							0.0	0.0	0.0
	Developing								
IV.	countries				17.1	28.9	29.2	38.5	42.3
	of which:								
	Α.	Asia			14.4	23.0	22.5	30.1	31.5
		a)		SAARC	2.9	5.4	4.3	5.4	5.7
				Other					
	_	b)		Asian		17.6	18.2	24.7	25.8
	В.	Africa			2.2	4.8	4.4	5.5	7.6
		Latin							
	<u>_</u>	American			0.5	10	2.2	20	2.0
14	Othors / una	countries			0.5	1.2	∠.3	3.0	3.2
٧.		pecilieu			2.9	1.5	4.3	0.3	0.4
	i otal i rade				100.0	100.0	100.0	100.0	100.0

 Table 4: Share of Region/Country in India's Exports: 1990-91 to 2007-08

Source: Estimated from RBI "Handbook of Statistics on Indian Economy", Directorate General of Commercial Intelligence and Statistics.

## 2.2 Trends in India's Services Exports

#### 2.2.1 Growth in India's Services Exports Overtime

In less than two decades, India has become one of the top five exporters of services amongst developing countries, and has surpassed some the other Asian countries that had dominated the services trade in the 1990s. India has been deemed as a major exporter of services in the world with a market share of 2.6 % in 2007 as against 0.6 % in 1995. India's services sector has matured considerably during the last few years and has been globally recognized for its high growth and development. Indian services exports grew at a compounded annual growth rate (CAGR) of 17% during 1993-2000 but have grown at a much faster pace recording CAGR of about 24% during 2001-2008. There has been rapid growth in the services exports from the year 2002. The exports have grown from US \$ 20.8 billion in 2002 to US \$ 90.1 billion in 2008.

U.S. is one of the major markets for export of services for India. Its share in total services exports has been around 10% with the growth of services exports to U.S. being higher than that to the world since 2005-06 (Figure 6).



Figure 6: Growth of India's Exports of Services to World and U.S.: 2002-2008

Slowdown in U.S. has led to lower growth of services exports to U.S. as well as to the world. India's export of services to U.S. grew at a rate of 76.2% in 2005-06 as compared to the earlier year, but declined to 34% and further to 31% in 2006-07 and 2007-08. Growth of exports of services to the world has declined marginally from 28% to 22% in this period. Interestingly, the share of U.S. in India's exports of services has not changed much overtime. Its share increased from 10.7% in 2006-07 to 11.6% in 2007-08 (Table 5).

	Export of Total Services (US Millions)	Export of Services to U.S.(US Millions)	Share of US in Total Exports of Services (%)
2000-01	16,268	1,955	12.0
2001-02	17,140	1,692	9.9
2002-03	20,763	1,875	9.0
2003-04	26,868	2,212	8.2
2004-05	43,249	3,359	7.8
2005-06	57,659	5,917	10.3
2006-07	73,780	7,919	10.7
2007-08	90,077	10,443	11.6

Та	b	le	5:	In	dia	's	T	otal	Ex	por	ts	of	Ser	vices	to	W	orld	and	U.S	5.
----	---	----	----	----	-----	----	---	------	----	-----	----	----	-----	-------	----	---	------	-----	-----	----

Source: Bureau of Economic Analysis and RBI

#### 2.2.2. Composition of India's Services Exports

India's export basket has not diversified very much overtime as around 40% of India's exports comprise exports of software services since 2000-01. Export of software services has grown at a compound rate of growth of 26% as compared to 24% of total services (Table 6). Apart from software services, travel and transportation services constitutes the export basket, with a share of around 12% and 11% respectively in 2007-08 (Figure).

		Invi	sibles by S	orvico Evo	ort of Tran	eactions						
	CAGR 1993 2000 01 2001 02 2002 02 2002 04 2004 05 2005 07 2007 08 CA											
	2000	2000-01	2001-02	2002-03	2003-04	2004-05	2005-06	2006-07	2007-08	2008		
Travel	4.56%	3,497	3,137	3,312	5,037	6,666	7,853	9,123	11,349	15.85%		
YoY Growth		15.18%	-10.29%	5.58%	52.08%	32.34%	17.81%	16.17%	24.40%			
Transportation	2.53%	2,046	2,161	2,536	3,207	4,683	6,325	7,974	10,014	21.96%		
YoY Growth		19.86%	5.62%	17.35%	26.46%	46.02%	35.06%	26.07%	25.58%			
Insurance	9.29%	270	288	369	419	870	1,062	1,195	1,639	25.29%		
YoY Growth		16.88%	6.67%	28.13%	13.55%	107.64%	22.07%	12.52%	37.15%			
G.N.I.E	52.75%	651	518	293	240	401	314	253	330	-8.14%		
YoY Growth		11.86%	-20.43%	-43.44%	-18.09%	67.08%	-21.70%	-19.43%	30.43%			
Miscellaneous of which:	31.99%	9,804	11,036	14,253	17,965	30,629	42,105	55,235	66,745	27.09%		
YoY Growth		-3.44%	12.57%	29.15%	26.04%	70.49%	37.47%	31.18%	20.84%			
Software		6341	7556	9600	12800	17700	23600	31300	40,300	26.01%		
YoY Growth			19.16%	27.05%	33.33%	38.28%	33.33%	32.63%	28.75%			
Total	16.91%	16,268	17,140	20,763	26,868	43,249	57,659	73,780	90,077	23.85%		
YoY Growth		3.56%	5.36%	21.14%	29.40%	60.97%	33.32%	27.96%	22.09%			

 Table 6: Composition of India's Exports of Services

\*G.N.I.E: Govt. Services not included elsewhere, Figures in million U.S. \$ Source: <u>www.rbi.org.in</u>

The major drivers of sustained year-on-year growth rates registered by aggregate Indian exportable services have been earnings from Travel, Transportation and Miscellaneous Services which accounts for both Software and Non-Software services.

- <u>Travel</u>, which is represented by the Foreign Tourist Arrivals and Foreign Exchange Earnings, registered a higher year-on-year growth rate of 24.40% in 2007-08 as compared to the previous year growth rate of 16.17%. Foreign Tourist Arrivals during the year 2008 were 5.37 million as compared to Foreign Tourist Arrivals of 5.08 million during the year 2007. Foreign Exchange Earnings (FEE) in US\$ terms during the year 2008 were US\$ 11,747 million as compared to US\$ 10,729 million in 2007. During April-September 2008, Travel Services registered a 22% growth rate as compared to 24% in the same period a year ago. However, the impact of global financial meltdown is evident in the latest numbers released by the ministry of tourism, India which reports foreign tourist arrivals at 1.461 million in 4Q 2008-09 were 13.75% lower as compared to 1.694 million in 4Q 2007-08. Also, foreign exchange earnings during the same period were lower at US\$ 2,731 million as compared to US\$ 3,935 million during January to March 2008.
- Export of Transportation services have slowed down in past few years registering 25.58% year-on-year growth in 2007-08 as compared to a growth rate of 46.02% in 2004-05 and 26.07 in 2006-07. Transportation was the only service recording a higher growth rate of 38% in April-September 2008 from a 10% growth rate in April-September 2007.
- <u>Insurance Services</u> registered a higher year-on year growth rate of 37.15% over the previous year growth rate of 12.52%. During April-September 2008, Insurance Services observed a meagre 1% growth rate as compared to a 29% growth rate in April-September 2007.
- <u>Non-Software services</u>, under Miscellaneous receipts, recorded a fall in year-on-year growth rate from 29.34% in 2006-07 to 10.49% in 2007-08. Communication, business and financial services were the major contributors to the decline in non-software services. Though Communication and Financial services recorded positive growth rates in 2007-08, the growth rates were substantially lower than previous year growth rates and similarly the decline was also attributable to a major negative growth rate recorded in export of business services. This slowdown is the result of the banking financial services and insurance sector being at the core of global economic slowdown. However, services such as Construction, News Agency, Royalties, Copyrights and License Fees and Personal, Cultural Recreational services registered higher year-on-year growth rates in the non-software category.

Miscellaneous Receipts:Non-Software										
	2005-06	2006-07	2007-08							
Communication Services	1,575	2,099	2,436							
YoY Growth		33.27%	16.06%							
Construction	242	332	780							
YoY Growth		37.19%	134.94%							
Financial	1,209	2,913	3,085							
YoY Growth		140.94%	5.90%							
News Agency	185	334	643							
YoY Growth		80.54%	92.51%							
Royalties, copyrights and license fees	191	97	157							
YoY Growth		-49.21%	61.86%							
Business Services	9307	19266	16624							
YoY Growth		107.01%	-13.71%							
Personal, Cultural, Recreational	189	173	559							
YoY Growth		-8.47%	223.12%							
Others	5607	1042	335							
Total	18,505	26,256	24,619							
YoY Growth		41.89%	-6.23%							
Source: www.rbi.org.in										

 Amongst the <u>export of business services-</u> Business & Management Consultancy as well as Architectural, Engineering and Other Technical Services registered the largest decline. Trade Related services recorded a substantial increase of 137% from the previous year.

Miscellaneous Receipts: Business Services				
	2005-06	2006-07	2007-08	
Trade Related	521	939	2,223	
YoY Growth		80.23%	136.74%	
Business and Management Consultancy	2,320	7,346	4,215	
YoY Growth		216.64%	-42.62%	
Architectural, Engineering and Other Technical	3,193	6,134	3,287	
YoY Growth		92.11%	-46.41%	
Maintenance of Offices	1,577	2,334	2,867	
YoY Growth		48.00%	22.84%	
Others	1696	2513	4032	
YoY Growth		48.17%	60.45%	
Total	9,307	19,266	16,624	
YoY Growth		107.01%	-13.71%	

Source: <u>www.rbi.org.in</u>

The Non-Software category recorded a 16% growth rate in April-September 2008 from the corresponding period a year ago. All services Under the Non-Software category, recorded positive growth rates but of importance were the Construction and Personal, Cultural & Recreational services registering the highest year-on-year growth rate of 45% and 52% respectively. Business Services recorded a moderate 14 % year-on-year growth rate in April-September 2008.

 Under Miscellaneous receipt - the export of Software Services has been a major contributor to the growth of exportable service accounting for 45% of total services export in 2007-08. During April-September 2008, Software receipts stood at US \$ 21.9 billion, showed a lower growth of 22.3 per cent than that of 26.3 per cent in same period a year ago. It should be pointed out that cost cutting becomes a top priority in times of the current economic deterioration which could mean a reduction in IT spending by advanced economies with negative implications for the growth of Indian Software Exports.

• In addition, Banking, Financial and Insurance (BFSI) sector which has been the epicenter of this global financial crisis accounts for approximately 50% of the revenues of IT & ITeS providers which makes IT & ITeS highly vulnerable to the current global slowdown in terms of delayed decision making and reduction in IT spending by customers of frontline IT companies.

#### 2.2.3 Direction of India's Exports of Services

Exports of services from India have been oriented mostly towards the EU25 and US in the developed world. India's country-wise exports of services show that the US and the UK are two most important destinations for service exports. EU and South East Asia are relatively less important destinations. According to the Economic Survey 2007-08, India exports travel services mainly to EU and transportation services to South East Asia.

Around 13% of the total Indian services exports were oriented towards the EU25 in 2003. However, the share has come down to 10% in 2005. US accounted for about 8.7% of total India's services export in 2005. Interestingly, the share of US has gone up to around 10.7% in 2007 (Table 7).

Year	Exports to US (\$ mn)	Share of US in Total Exports (%)
2003	2000	7.4
2004	2886	6.7
2005	5057	8.8
2006	7693	10.4
2007	9664	10.7
2008	12141	-

 Table 7: Services Exports of US and Share in Global Indian Services Exports

Source: Bureau of Economic Analysis

Though, the impact of global slowdown on India's exports of services has not been as deep as the impact on goods and services exports are still recording positive export growth, the increasing legislations and inbuilt conditions in the stimulus packages offered for revival in the developed countries may lead to esclating the impact of slowdown on services exports overtime.

For example, under the American Recovery and Reinvestment Act (ARRA), 2009, the U.S. government has restricted the companies availing bailout package from replacing American laid off workers with low-cost H1 B visa professionals. Currently, the cap for H1 B visa holders stands at 65,000 a year, of which approximately 40,000-45,000 holders

are IT professionals of Indian origin. This will have an adverse impact on services exports under GATS Mode 4.

Similarly, the British government has raised minimum requirement to enter Britain under the Tier 1 category from a graduate degree with a minimum salary of £17,000 to a master's degree with a minimum salary of £20,000. Ban by the U.S. and restrictive employment policy by the U.K. will on an aggregate level affect Indian export of services under GATS mode 4 and possibly restrain the new employment generation for the Indian IT and ITeS sector.

## 2.3. Trends in India's Imports of Goods and Services

Since 2001-02 onwards, India's merchandise imports have always been higher than its merchandise exports, leading to a negative trade balance which has grown over the years (Figure 7). Not only are the imports higher than the exports, they are also growing at a much higher rate. In 2008, India's exports grew by 23.7% while its imports grew by 38%.



Figure 7: India's Exports, Imports and Trade Balance: 2000-01 to 2007-08

In terms of services, however, export growth is much stronger than import growth which has led to an ever-growing positive trade balance in India's services trade. This reflects the importance of services sector in India's total trade.

Within merchandise imports, it is the oil imports of India which is much higher than the non-oil imports.

## • <u>Oil imports</u>

Since October 2007, there has been a steady rise in imports of oil. However, much of this increase can be attributed to increase in oil prices. After July 2008, there has been a drastic decline in India's oil imports on account of fall in prices. Volumes of Oil imports grew at almost 212 % in 2008-09 over 2004-05 (Table 8). The rate of growth of oil imports in each financial year over previous financial year remained greater than 30% except in 2008-09 (17%).

FY	Oil imports (\$ millions)	ROG (%)	
2004-05	29,844		
2005-06	43,963	47.31	
2006-07	57,099	29.88	
2007-08	79,715	39.60	
2008-09	93,176	16.88	

 Table 8: India's Oil imports and Rates of growth (%).

#### • Non-oil imports

India's Non-Oil imports have increased steadily over time (Table 9). Non-oil imports grew at almost 138 % in 2008-09 over 2004-05. However, in 2008-09 the growth rate fell from 33.8% in 2007-08 to 13.16%.

 Table 9: India's Non-Oil imports and Rates of growth (%).

FY	Non-oil imports (\$ millions)	ROG (%)
2004-05	81,673	
2005-06	105,203	28.81
2006-07	128,505	22.15
2007-08	171,940	33.8
2008-09	194,584	13.16

Within the import basket, composition of imports between oil and non-oil imports does not seem to have changed much overtime for India (Figure 8).

Figure 8: Composition of India's Import Basket: 2004-05 to 2008-09



#### **Conclusions from Trends**

- 1. Financial crisis in U.S. which began in the second quarter of 2007, adversely affected India's merchandise exports to U.S. with negative effects becoming pronounced from October 2008 onwards. A likely explanation for this lag is that India's exports grew at a much higher rate to the world as compared to US since 2005. Also, the share of U.S. in India's exports has declined over the years which reduced dependence of India's exports on U.S. market.
- 2. India's export basket in terms of its composition has diversified over time and shares of traditional exports has declined in the export basket. This has led to reduced dependence on few exportable products and helped moderate the impact of reduced demand of exports. However, there exists large scope for further diversification.
- 3. Overtime, the significance of South-South trade for India is increasing with the share of developing countries increasing from 17% in 1990-91 to 42% in 2007-08. In particular, the direction of India's exports is slowly shifting towards Asian developing countries. However, developed countries, like EU and US, are still India's major export markets.
- 4. In terms of exports of services, there has been an exponential rise overtime with CAGR of about 24% during 2001-2008. U.S. remains the major export market for India's services and software exports remain the major exportable service with 40% share.
- 5. The major drivers of sustained year-on-year growth rates registered by aggregate Indian exportable services have been earnings from Travel, Transportation and Miscellaneous Services which accounts for both Software and Non-Software services. Growth rate of exports has drastically fallen in all these services since 2007-2008 but has remained positive.
- 6. India's import growth has declined during the slowdown but the decline has been lower than the decline in exports. Non-oil import growth declined from 29% in 2005-06 to 13% in 2008-09 while oil imports declined from 47% in 2005-06 to 17% in 2008-09. Unlike growth rate of exports, the growth rate of imports has remained positive

The trends in India's exports and imports indicate that the impact of slowdown on India was felt with a lag probably due to overtime diversification in India's exports both in terms of composition and direction. <u>However, there is a large scope for further diversification both in terms of composition and direction of exports</u>. Around 30% of exports are still directed towards developed countries, which need to be diversified to developing countries. Share of fewer commodities in top 50% of India's exports at six digit level in 2007 as compared to earlier period reflects the need and scope for further diversification.

#### 3. Impact of Slowdown on India's Exports

Global demand plays an important role in determining export growth of a product. With rise in global incomes, demand for normal and luxury products rise while for inferior products it may decline. Income elasticity of demand<sup>6</sup> for luxury products is expected to be greater than one, while for normal goods it is expected to be between 0-1. The kind of products a country exports, i.e., income elasticity of demand of the product, is an important factor which determines the impact of slowdown on the country's exports. Along with income elasticity, price competitiveness may also determine the impact of slowdown on exports. If the products exported are less price sensitive, then in case of a slowdown the option of lowering prices to maintain market shares may not be feasible.

Econometric estimation of the price and income elasticity of imports has been the subject of a large literature<sup>7</sup> both for developed and developing countries. Apart from price competitiveness, many other factors may affect demand for a product, e.g. income of consumers, tastes and preference, etc. Income elasticities of demand are said to capture market sensitivity to non-price factors (Fagerberg, 1988 and Meliciani 2001). Most empirical studies find that the exports of developing countries, especially in Asia, have low price elasticities but high income elasticities (Goldstein and Khan, 1985; Marques and McNeilly, 1988; Feenstra, 1994, Senhadji and Montenegro, 1999).

The empirical evidence of low price elasticity and high income elasticity of export demand in general has important implications for exports of developing countries. Firstly, this suggests that the export growth of developing countries is highly dependent on the economic performance of developed countries. Secondly, it implies that the developing countries may have limited feasibility of using price competition to maintain or increase exports.

It has been recognized in the literature that the higher the income elasticity of the export demand, the more powerful will exports be as an engine of growth<sup>8</sup>. Senhadji and Montenegro (1999) found that the Asian countries had the highest estimated values for income elasticity among the developing and industrial countries. This advocated the view that exports had been a powerful engine of growth in the Asian region. This has an important implication: the higher the income elasticity of export demand the more severe will be the impact of slowdown of incomes/GDP on developing countries exports and growth.

To estimate the extent of impact of slowdown of global GDP growth on India's exports, the study estimates income elasticity of export demand for India's total exports and its

<sup>&</sup>lt;sup>6</sup> the income <u>elasticity</u> of <u>demand</u> measures the responsiveness of the demand of a good to the change in the income of the people demanding the good. It is calculated as the ratio of the percent change in demand to the percent change in income. For example, if, in response to a 10% increase in income, the demand of a good increased by 20%, the income elasticity of demand would be 20%/10% = 2.

<sup>&</sup>lt;sup>7</sup> see, for example, Malley and Moutos (2002), Erkel-Rousse and Mirza (2002), Caporale and Chui (1999), Hooper et. al. (1998)

<sup>&</sup>lt;sup>8</sup> See Huuthaskker and Magee (1969), Goldstein and Khan (1985)

sectoral components. The elasticities indicate the extent to which India's exports will increase/decrease in response to changes in global demand captured by changes in global GDP growth. These income elasticities are then used with GDP growth forecasts for 2009 and 2010 (provided by OECD Economic Outlook, March 2009) to arrive at the estimated impact on India's total and sectoral export growth to world.

### 3.1 Methodology and Data

For assessing the impact of slowdown on India's exports, we estimate the standard export demand equation for India using data for 1970 to 2008. According to the standard export demand function, exports depend on price competitiveness as measured by the real exchange rate and global income as measured by global GDP. For India, many of the tradables comprise low-technology products, such as leather footwear, gems and jewellery, marine products, etc. therefore there is a strong possibility of these being highly differentiated products, with close substitutes available. Demand for these products is therefore expected to be price-sensitive. Firms offering a lower relative price would be able to sell more than their competitors.

To measure relative price, it is necessary to look at price and exchange rate data. The volume of exports depends on nominal exchange rates after adjusting for the domestic level of inflation<sup>9</sup> by which we arrive at real effective exchange rate (REER). While considering exports, a country's REER would preferably reflect not only its price competitivess vis-à-vis the importing country but also its price competitiveness versus competing exporters to the same country. In other words, relative exchange rate index construction for exports involves the added complication of taking third party competition into account. This approach has been followed by a number of studies (e.g. Spilimbergo et al. 2003, Wijeweera et al 2008).

To capture the relative difference in international and domestic market prices, a ratio of world GDP deflator to India's GDP deflator is used. Real exports are arrived at by deflating nominal exports with export unit value index (source: Reserve Bank of India). World GDP in real terms captures the income effect. This is a standard proxy for capturing income effect. The model estimated is therefore as follows:

Where  $LNEXP_{INDIA}$  is log of real exports of India to the world;  $LNGDP_{WORLD}$  is log of real world GDP; and LNREER is a product of effective exchange rate and relative prices proxied by ratio of world GDP deflators and India's GDP deflator. The data for world GDP at current and constant prices is taken from World Development Indicators;

<sup>&</sup>lt;sup>9</sup> Real exchange rate (R) = nominal exchange rate (e)  $\times$  foreign price (p\*)/domestic price (p). The nominal exchange rate is measured as domestic currencies per unit of foreign currency.

exchange rate is taken from ERS International Macroeconomic Data Set; and India's merchandise export is taken from World Integrated Solutions (WITS; COMTRADE).

Empirical evidence suggests that India's exports react favourably to devaluation or depreciation. Following the devaluation of rupee in 1991 there was a spurt in export growth. Studies have reported that price competitiveness of India's exports is an important determinant of the volume of exports and that rupee depreciation can have a significant positive effect on its current account balance (Joshi and Little, 1994; Srinivasan 1996; Banik 1999). It is therefore expected that price elasticity given by  $\alpha_3$ 

will be negative.

Apart from relative prices, the global GDP is also considered to be an important variable for estimating export demand functions. As stated above, many studied have found income elasticity, which is given by coefficient of LNGDP <sub>WORLDt</sub> i.e.,  $\alpha_2$  will be positive.

We have followed the standard procedure in the literature to check for unit roots in each series before estimating a model that involves time series data. If there is a unit root, then that series is considered to be non-stationary. The stationarity of each series is tested by the following unit root tests: (a) Augmented Dickey-Fuller test (ADF test); and (b) the Phillips-Perron test (PP test). Since regressions have been run for aggregate exports as well as sector-specific exports, we have undertaken tests separately. The results of these are reported in the Annex I. We find that most of the series used are stationary at levels. Wherever, we found that the series contains the unit root in levels, but no unit roots in first differences, we have used the popular Engle and Granger (1987) method to estimate the export demand functions. According to Engle and Granger (1987), it is possible to have a linear combination of these non-stationary variables that is stationary. Two estimation steps are carried out. First, the best possible linear equation - as shown in equation (1) - is estimated and residuals are collected. Then a unit root test is used to test whether residuals are stationary. We find that they are stationary, which implies that there exists a long-run equilibrium relationship and therefore a meaningful regression estimate can be carried out.

## 3.2 India's Income Elasticity of Total Exports

To estimate the impact of slowdown of world GDP growth on India's export growth, we estimate the above equation (equation 1). Similar equations have been estimated to arrive at price and income elasticities of India's exports to US, G7 countries, advanced economies (as defined by IMF) and ASEAN 5. Results are reported in Table 10.

Countries	Price Elasiticity for	Income demand	
	Exports	elasticity for Exports	
World	-0.54*	1.88*	
G 7 countries	-0.21*	1.06*	
USA	-0.36*	2.48*	
ASEAN - 5	-0.42*	1.11*	

Table 10:	Income	Elasticities	of India	's E	xport I	)emand
I GOIC IV.	meome	Linguicitics	or mana			/ unu

Note: G7 countries are as per the IMF definition. \* denotes significant at 1%.

The results show that India's exports to world are much more responsive to income changes as compared to price changes, though both the factors are found to be significant. A 1% decline in GDP growth of world will lead to 1.88% decline in India's growth of exports to world. However, a much higher price competitiveness is required to increase exports. It should be noted that the price elasticity inter-alias captures the effect of depreciation of the currency and lowering of relative prices. This implies that to compensate for the loss in export growth, it will be very difficult to increase India's export growth through improvements in its price competitiveness. A 10% reduction in prices will lead to 5.4% increase in exports.

Income elasticity of India's exports is found to be highest with respect to U.S., i.e. 2.48 which implies that a slowdown in U.S. with respect to GDP growth will have a more significant impact on India's export to U.S. as compared to a decline in growth of world GDP. The income elasticity with respect to ASEAN 5 countries is found to be 1.11 which is comparatively lower than income elasticity of India's exports with respect to world and U.S. This implies that a slowdown in growth of ASEAN GDP will have a lower impact on export growth of India to ASEAN 5 as compared to the world.

India's exports are found to be more price elastic with respect to ASEAN 5 as compared to US and G7. This indicates that with respect to ASEAN 5, India is exporting much more differentiated products with close substitutes as compared to other developed countries. Though earlier studies have found a much higher price elasticity for India (e.g. Srinivasan 1996), more recent studies have found lower price elasticity (e.g., Banik 2008). An apparent reason for this is a change in the composition of India's exports from price-sensitive items to less price-sensitive items such as chemicals, engineering goods and petroleum products. An important implication of this is that slowdown in ASEAN 5 countries may have less adverse impact of India's export growth, therefore exploring further export opportunities in these countries could be considered.

#### 3.3 India's Income Elasticity of Sectoral Exports

Following similar methodology outlined in the earlier section, income and price elasticities are estimated for ten major sectors of export of India to the world. Detailed results with respect to stationarity of the series and other test statistics are presented in the Annex I. The price and income elasticities are reported in Table 11.

		Price Elasticity (1)	Income Elasticity (2)
1	Textiles and Textile Products	-0.29*	1.16*
2	Ore & minerals	-1.27*	4.85*
3	Leather and Leather products	-0.88*	1.25*
4	Marine Products	-0.47*	1.26*
5	Plantation	-1.05*	0.33*
6	Chemicals Chemical products	-0.23	2.55*
7	Petroleum products	-1.30	5.40*
8	Engineering and Electronic Products	-0.56*	2.28*
9	Agriculture and allied Products	-0.71*	1.38*
10	Gems & Jewellery	-0.92*	4.11*
	Total exportS	-0.54*	1.88*

 Table 11: Price and Income Elasticities for India's Major Sectors of Exports

Note: \* denotes significant at 1%.

The results show that income elasticity of total exports of India is very high, i.e., 1.88. Estimates of income elasticities of ten major export sectors of India (which are around 95% of total India's exports) show that sectors such as petroleum products, ores and minerals, gems and jewellery, chemical products and engineering products is high. India's traditional export sectors like textiles leather and plantation have relatively low income elasticity, with lowest being for plantation. This also explains the shift in India's exports away from traditional exports and growing diversification of export basket in the period 2000-2007 in which global GDP grew consistently.

Two observations can be made here. *First*, India's exports of textiles is not a high value added exports since as the incomes of the people rise their demand for India's textiles does not grow as significantly as their demand for other products exported by India. Improvement in brand name and quality is needed for increasing the income demand elasticity for textiles. The same is true for leather and leather products. For both textiles and leather exports, price elasticity is low which implies that improving cost competitiveness or lowering prices may not be a feasible option for boosting exports of these sectors during slowdown.

*Second*, exports of products like plantation are not expected to be income elastic as their demand may not be linked to incomes of the people. However, price elasticity is found to be very high for plantation which implies that lowering of their prices and improving their cost competitiveness can boost their exports in the period of slowdown. High price elasticities are also found for ores and minerals. Lowering their prices to boost exports in the period of slowdown can be considered as a mitigating step. But price elasticities are not found to be a significant factor in export growth for petroleum and chemical products and these are the products which are rising in their share in India's total exports.

An important implication of these elasticities is that during slowdown in growth of global GDP, sectors with higher income elasticities will experience a higher decline in their export growth. But if the price elasticities are also high then these sectors can lower their prices to improve exports, but such an option may not be available to products with high income elasticity but low price elasticity like gems and jewellery and textiles. These sectors are relatively more vulnerable sectors of the economy in terms of impact of global slowdown.
#### 4. Forecast of India's Exports using Income Elasticity of Exports

Using the income elasticities for export demand for India and the forecasted change in global GDP growth, India's total export growth and sectoral export growth for ten major sectors has been estimated for the year 2009-10 and from 2010 April to 2010 December. The forecasted slowdown of GDP growth as provided by OECD Economic Outlook (March 2009) are used. The slowdown in GDP growth projected by OECD are as follow:

	2006	2007	2008	2009	2010
World	4.3	4.1	2.2	-2.7	1.2
USA	2.8	2	1.1	-4	0
Euro Area	3	2.6	0.7	-4.1	-0.3
Japan	2	2.4	-0.6	-6.6	-0.5

<b>Table 12: Projected Real GDP</b>	Growth (%): 2009 a	nd 2010
-------------------------------------	--------------------	---------

Source: OECD Economic outlook (March 2009)

According to the projections, global GDP growth is expected to decline from 2.2% in 2008 to -2.7% in 2009 but it is expected to revive in 2010 to 1.2%. However, positive GDP growth is not forecasted for developed countries like USA, Euro Area and Japan in 2010.

The results of the estimates are presented in Table 13. The results show that total export will grow by -2.2% in 2009-10, which implies that there will be almost flat growth marginally tending towards a negative growth. Most of the sectors experience a negative growth rate. Positive growth in exports is forecasted for plantation, agriculture sector and engineering & electronics sector. It should be noted that though positive growth rates of exports in agricultural products has been forecasted, it is much lower than the 55% export growth in 2007-08.

	Export	Export	Export	Projected
	growth	Growth 2008-	Growth 2009-	Export
	in 2007-	09 over 2007-	10 over 2008-	Growth 2010
	08	08	09	over 2009
Textiles and Textile	15.7	-8.9	-3.6	4.6
Products				
Ore & minerals	30.4	-12.3	-4.9	26.6
Leather and Leather	16.3	2.5	-1.6	5.7
products				
Marine Products	-2.6	-4.4	-0.1	5.3
Agriculture	55.6	2.6	1.5	14.6
Plantation	11.6	54.6	14.2	14.3

Table 13: Forecasted Total Merchandise Export Growth Sectoral Export Growths:2008-09 and 2009-2010

Engineering & Electronics	26.6	22.0	0.4	9.5
Chemicals & Products	21.5	9.7	-4.3	8.1
Gems & jewellery	23.3	-4.9	-11.1	15.3
Petroleum products	52.0	4.7	-11.8	21.2
Total sectors	29.1	3.40	-2.2	8.3

The forecasts show that petroleum products will experience the maximum decline in export followed by gems and jewellery, ores and minerals and textiles and textile products. Total exports have grown by 3.4% in 2008-09, declining from 29.1% growth in 2007-08. The predicted export growth in 2009-2010 is -2.2%, which is predicted to increase to 8.3% in 2010-11 (April to December). The estimates show that all sectors experience a positive growth in exports if the global GDP growth is positive, as predicted by OECD.

However, although many sectors show a positive export growth rate in 2008-09, a close examination of quarterly trends reveal that this positive export growth masks the decline in export growth during October 2008-March 2009. For example, sectors such as leather and leather products and petroleum products experienced a negative export growth of 10% and 28%. In 2010-11, sectors such as agricultural products, plantation, engineering, chemicals and petroleum products are expected to reach initial level of exports in 2007-08.

#### 5. Impact of Slowdown on Employment through International Trade

The predicted overall export growth for the years 2009-2010 and 2010-11 and sectoral export growths have been used to estimate the impact of global slowdown on employment in the economy. The methodology adopted for this is described in the Box below.

#### **BOX: Methodology for Estimating Impact on Employment**

Using the latest available input-output matrix for India for the years 2003-04, impact of predicted change in exports on employment has been estimated for 10 major sectors of the Indian economy for the years 2008-09, 2009-2010 and 2010 (April to December).

Using the actual sector-wise exports for the years 2006-07 and 2007-08, provided by RBI, change in exports has been calculated for sub-sectors of input-output matrix. Using Leontif inverse matrix, the change in output across different sectors consequent to change in output for each sector (due to change in exports) has been estimated. Applying the labour coefficients across the sectors, total employment change (which is direct as well as indirect) is arrived at for each sector. These are further summed up to arrive at change in total employment and change in employment for ten major sectors.

The estimated impact on employment for a sector includes both direct increase in employment of the sector caused by exports as well as indirect increase in employment which is generated because of the rise in exports of other sectors which use the sector's output as inputs. For example, employment in agricultural products may rise because of increase in their exports and also because of increase in demand for their products as exports of processed food products and textiles and textile products increase.

The results are presented in Table 14. The estimates show that in the year 2008-09, with export growth of 3.4%, the total job loss in India due to lower export growth was of around 1.16 million. However, since the impact of slowdown on India's exports was strongly felt only after September 2008, the net employment created by exports in this year was positive, i.e., 1.25 million. The net employment is sum total of jobs created and lost in different sectors overtime. In the year 2009-10, export growth is predicted to be - 2.2%, and the total job loss is estimated to be around 1.3 million. However, since export growth is positive for some sectors like plantation and these sectors have high employment multipliers, the net employment loss is estimated to be -748 thousand.

For the year 2010-11, estimation could be done only for three quarters, i.e., till December 2010 as GDP growth predictions are not available beyond that. Using the predicted export growth of 8.3%, the total employment generated in the economy is estimated to be

5.22 million. No job losses are expected as all sectors are expected to experience positive export growth.

Sector-specific employment changes are reported in Table 14. In 2008-09, job losses are likely to arise in sectors with negative export growth like textiles and textile products, ores and minerals, marine products and gems and jewellery. In 2009-2010, most of the sectors are predicted to have job losses, except for agriculture and plantation for which positive export growth has been predicted. Maximum job losses are likely to occur in gems and jewellery sector followed by ores and minerals, textiles and textile products, and petroleum products. For the year 2010-11 (till December), we find that there is employment generated due to change in exports in all sectors.

	Employment	Employment	Employment
	projection in	projection in	projection in 2010-11
	2008-09	2009-10	(Till December 2010)
Ores and Minerals	-373,023	-440,961	936,824
Textiles&Products	-559,621	-253,810	260,172
Leather & Products	30,787	-21,102	54,784
Marine Products	-16,498	-96	16,484
Agriculture	373,148	159,070	2,468,094
Plantation	1,275,376	422,672	561,494
Engineering and	665,445	-24,927	332,997
Electronics			
Chemicals& Products	45,114	-29,856	49,504
Gems & Jewellery	-217,151	-505,023	465,005
Petroleum Products	33,749	-54,045	79,445
Net Employment	1,257,327	-748,078	5,224,802
Job Loss	-1,166,293	-1,329,820	

 Table 14: Impact of Slowdown on Employment: 2008-09 to 2010-11

During 2010-11 (till December 2010) with all sectors likely to experience positive export growth, the declining trend in employment would be reversed. In sectors such as ore and minerals, leather and leather products, engineering products, chemical and petroleum products, the additional employment generated due to export growth in 2010-11 is likely to compensate for job losses due to decline in exports during the preceding two years. However, in respect of textile sector and gems and jewellery the export growth during the nine months of 2010-11 would not be sufficiently buoyant so as to compensate for job losses during the preceding two year.

#### 6. Identification of Sectors for Employment Generation

One of the immediate policy actions which may be required to mitigate the impact of global slowdown on employment is to identify the export sectors which have large employment multipliers. An employment multiplier of a sector gives an estimate of aggregate direct and indirect employment changes (in person years) resulting from increase in one unit of output of the sector. The indirect employment changes occur due backward and forward linkages of the sector in the economy. Thus, employment multipliers will indicate the extent of economy-wide employment generated.

Using the latest available input-output matrix, employment multipliers have been generated for the ten major sectors, which are as follows.

	Employment multiplier
Plantation	2.15
Agriculture	3.20
Marine Products	0.56
Ores and Minerals	2.00
Leather & Products	1.11
Gems & Jewellery	0.50
Textiles & Products	1.22
Chemicals & Products	0.36
Engineering and Electronics	0.38
Petroleum Products	0.26

 Table 15: Employment Multipliers based on input-Output Matrix of 2004-04.

As seen in Table 15, the employment multipliers for agricultural sector is highest-3.2, followed by ores and minerals. However, the employment generated in the economy will depend on value of exports of the sector and its employment multiplier. <u>Sectors with high employment multipliers which may not be able to regain their export growth to the initial level of 2007-08 in 2010 are – textiles & products (1.22); leather & products (1.11); and gems and jewellery (0.5).</u>

### 7. Conclusions and Mitigating Strategies

Riding on the back of brisk growth in the global economy since 2002, India's exports witnessed a phenomenal three-fold increase during the period 2002-03 and 2007-08. This powerful dynamo for employment generation is now threatened by rapid contraction in global demand and weakening labour market. It is a major challenge for India to properly manage the fallout from the current global slowdown on its export sector and limit the adverse consequences for employment situation in the country.

As estimated in this paper, India's export growth during 2009-10 over the previous year is likely to be flat, tending towards the negative side (-2.2%). However, in conjunction with recovery in demand in developed economies, India's export prospects are likely to improve during the period 2010-11. Exports in sectors such as textiles and clothing; ores; marine; and gems and jewelry are likely to decline significantly during 2009-10, as compared to the previous financial year. Relatively high employment – output multipliers in these sectors is likely to result in high job losses. The position of export-related employment is likely to improve during April-December 2010, due to an improved export performance in sectors such as chemicals; petroleum products; and engineering and electronic products. However, additional employment created due to export growth during this period in textiles and gems and jewellery will not compensate for job losses in these two sectors for the proceeding two years.

Overall, it is apprehended that the contribution of the export sector in generating employment in India is likely to remain under stress till 2009-2010 with improvements in the year 2010-11. The <u>net employment</u> loss is estimated to be around 748,000 in 2009-2010, with exports generating total employment of 5.2 million in the year 2010-11 (till third quarter).

To build resilience of the economy to trade shocks and improve competitiveness of the exports, it would be useful for the government to consider mitigating strategies. This study suggest five specific mitigating strategies relating to (a) diversification of exports to new geographical destinations and new products; (b) simplification in customs procedures for reducing transaction costs; (c) examination of the likely impact of antidumping and safeguard duties imposed by India on down-stream user industries; (d) measures aimed at assisting exporters to retain their market presence during the crisis period; and (e) expeditious multilateral examination of adverse impact of bailouts and stimulus packages and prompt remedies.

#### (a) Diversification of exports: identifying new markets and new products

Despite targeted efforts by the government for seeking new geographical destinations for India's exports, the European Union and the United States continue to be the main destination of India's exports. These two main markets account for nearly 30% of India's exports, although the share of the US in India's exports has reduced gradually over the years. While demand in most countries has been adversely affected by the current global

slowdown, the extent and timelines for recovery vary considerably. On the basis of available forecasts, it is likely that countries/ regions such as West Asia, ASEAN, Australia and Brazil are likely to witness a faster recovery than other economies. These countries can provide viable and sustainable alternate markets for reducing India's overwhelming reliance on the EU and the US for its exports. There is a need to develop and implement measures that would ensure sustained export growth which is not impeded by adverse developments in big foreign markets or in respect of few products.

In each of the importing destinations (ASEAN, Australia, Brazil, Korea, West Asia, and South Africa) competitiveness analysis of India, importing country and five main exporting countries has been undertaken for identifying products in which India has the potential to significantly increase its exports from the current level (potential products) or start exporting the new product<sup>10</sup>. Around 958 products identified. The list of potential products and new products in each of these import destinations is provided in ANNEX II. As shown in Table 16 India has the potential to increase its exports of new and potential products by almost USD 35 billion.

Country	Estimated Value of New Exports of India (in bn \$)	Estimated Value of Potential Products Exports of India (in bn \$)
Australia	1.98	2.19
Brazil	3.36	0.19
Korea	6.55	1.45
UAE	0.70	2.70
Malaysia	3.65	0.80
Philippines	2.20	1.05
Thailand	2.60	0.85
Singapore	0.85	2.52
South Africa	1.35	0.20
TOTAL	23.24	11.95

Table 16: Potential Gain for India from Export of New and Potential Products

As a first step in harnessing this potential, it may be useful for the industry and government to identify specific reasons why India's comparative advantage in these products has not translated into export gains. As India is in the process of negotiating free trade agreements with most of these countries, this opportunity could be used to address border and behind-the-border trade-related constraints identified in the importing country. Early conclusion of free trade agreement negotiations and implementation of the agreement with some of these countries could provide India with attractive markets for reducing the risk of overall exports being adversely affected by developments in a few big markets.

<sup>&</sup>lt;sup>10</sup> Bilateral and global RCAs (revealed comparative advantage) have been used for competitiveness analysis.

For India to again achieve export growth witnessed prior to the global slowdown, the need to preserve the existing market access in big economies becomes extremely important. While an early and satisfactory conclusion of Doha Round would help in this regard, it is also essential to be vigilant that non-tariff measures do not act as a disguised trade restriction.

#### (b) Simplification in customs procedures for reducing transaction costs

With profit margins shrinking globally, cost competitiveness would be an important determinant for retaining or acquiring a share in export markets. A part of the cost-cutting efforts are linked with government's initiatives aimed at facilitating trade. In an attempt to reduce some of the transaction costs associated with international trade, the government has been simplifying its customs procedures over the past few years. While this is a continuing process, it needs to gather significant additional momentum, if India's exporters are to cut costs further, enhance their competitiveness and retain or increase their market share in foreign markets. Using the costs and procedures involved in importing and exporting a standardized shipment of goods, World Bank's report *Doing* Business 2009 states that India has slipped 9 ranks in respect of trading across borders. This report suggests that considerable procedural improvements related to international trade remain to be undertaken as India's exporters require twice the number of documents for exports as compared to OECD countries. Similarly, the time required for export and import continues to be considerably higher in India compared to OECD countries. The possibility of further simplification, at least in these two areas in the short term, merits close attention of the government.

#### (c) Examining the likely impact of anti-dumping and safeguard duties on downstream user industries prior to imposition of the duties

While India has been a major user of anti-dumping measures over the past few years, there has been a significant increase in the number of fresh anti-dumping and safeguard investigations initiated from October 2008 onwards. Many of the products currently under investigation are chemicals and other intermediate products which are inputs for downstream industry. Imposition of anti-dumping and safeguard duties on products which are inputs for subsequent stage of industrial production, would increase the overall cost of production. The duties would also adversely affect export prospects, if the duties are imposed on imported inputs used for producing export-oriented goods. While the underlying anti-dumping and safeguard investigations are required to be undertaken in accordance with the requirement under relevant domestic law, the possibility of not imposing the duty on account of consumer interest does exist. In the context of current global slowdown, it may be beneficial for the economy as a whole if a detailed economic analysis on the likely impact of the duties on downstream user industry is undertaken, prior to imposition of the duties. In case the economic analysis estimates considerable increase in production costs, particularly of exports, the option of not imposing the duty on ground of consumer interest could be considered by the government. This would prevent the possibility of India's exports becoming uncompetitive on account of antidumping and safeguard duties on imported inputs.

# (d) Measures aimed at assisting exporters to retain their market presence during the crisis period

With economic recovery being predicted for 2010, it is important that India's exporters do not withdraw from the export market in the intervening period of down turn, if they are to take advantage of export opportunities during the period of recovery. Government could consider a two-pronged approach for supporting exporters to retain their presence in foreign markets. On the one hand it could support exporters through incentives such as easing trade financing. However, as export related incentives can be neutralized or offset by the importing country through imposition of countervailing duty, an attempt could be made at the multilateral level to explore the possibility of a stand still on countervailing duties that might otherwise arise from incentives given by developing countries. As an alternative, WTO's Subsidies Committee could consider the possibility of increasing the threshold level of subsidization below which no countervailing duty would be leviable. This option could represent a balance between the interests of exporters and domestic industry in the importing country.

# (e) Expeditious multilateral examination of adverse impact of bailouts and stimulus packages and prompt remedies

Many developed and some developing countries have implemented bailout and stimulus packages for countering the adverse impact of global slowdown and stimulating domestic demand. A possibility that some measures in these packages could adversely affect India's export interest exists. The government could consider putting into place a mechanism, at least in the short term, for constantly reviewing the implementation of these packages and identifying measures, if any, which has an adverse impact on India's interest. While India could consider resorting to WTO's dispute settlement mechanism for seeking redress against the identified measures, this is a time consuming process which could take up to 2 years and hence not likely to provide prompt relief to India. The government could consider a multilateral solution to this problem, whereby WTO members would agree that Subsidies Committee constitute a Group of Experts (comprising legal experts and economists), which would examine the complaint against specific measures in the bailout and stimulus packages and give its findings expeditiously, say within 3 months of the matter being referred to it. The specific measure would need to be modified or withdrawn promptly, if the Group of Experts finds that adverse effects have arisen due to the bailout and stimulus packages.

In addition to implementing the mitigation strategies outlined above, there is a need to develop and implement long term measures that would ensure sustained export growth which are not impeded by adverse developments in big foreign markets. The current global slowdown will have a silver lining if the opportunity offered to diversify exportable products and markets and enhancing competitiveness is fully utilized by the Indian industry.

#### **References:**

- Asian Development Bank. 2008. Special Note to the Asia Economic Monitor: Developing Asia's Prospects in the Global Slowdown.
- Banik, N. 1999. Indo-Sri Lankan investment opportunities under the new trade regime, Foreign Trade Review, vol. 34, No. 1, pp. 72-91.
- Caporale, G.M and M. Chui (1999). *Estimating income and price elasticities of trade in a cointegration framework*, Review of International Economics, 7, 254-264.
- Clarida, Richard H. 1993. Permanent Income, Import Prices, And The Demand For Imported Consumer Durables: A Structural Econometric Investigation, National Bureau Of Economic Research, working paper no. 4437, August.
- Dées and Vansteenkiste. 2007. The transmission of US cyclical developments to the rest of the world. Working Paper series No. 798, August 2007.
- Duarte, A. and Holden, K. 2003. *The Business Cycle in the G-7 Countries*. International Journal of Forecasting, vol.19, 685-700.
- Emran, M. Shahe and Shilpi, F. 2007. Estimating Import Demand Function in Developing Countries: A Structural Econometric Approach with Applications to India and Sri Lanka, Institute for International Economic Policy, Working Papers Series no.10, June.
- Fagerberg. J. 1988. International Competitiveness, The Economic Journal, 98 (June): 355-374.
- Feenstra, R., 1994. New Product Varieties and the Measurement of International Prices, American Economic Review, 84, 157-177.
- Garg, Deepak & Sandeep, Ramesh 2005. *Income and exchange rate elasticity of imports and exports*, working paper series, April.
- Giannone, D. and Reichlin, L. 2006. *Trends and cycles in the euro area: how much heterogeneity and should we worry about it?* Working Paper Series 595, European Central Bank.
- Goldstein, M., and M.S. Khan, 1985. Income and Price Effects in Foreign trade in R. Jones and P. Kenen (eds.), Handbook of International Economics, 2 (North-Holland, Amsterdam), 1041-1105.

- Hélène Erkel-Rousse & Daniel Mirza, 2002. Import price elasticities: reconsidering the evidence, Canadian Journal of Economics, Canadian Economics Association, vol. 35(2), pages 282-306, May.
- Houthakker, Hendrik S & Magee, Stephen P, 1969. *Income and Price Elasticities in World Trade*, The Review of Economics and Statistics, MIT Press, vol. 51(2), pages 111-25, May.
- IMF. 2008. *Financial Stress, Downturns, and Recoveries*. World Economic Outlook October 2008.
- Jim Malley & Thomas Moutos, 2002. Vertical product differentiation and the import demand function: theory and evidence, Canadian Journal of Economics, Canadian Economics Association, vol. 35(2), pages 257-281, May.
- Joshi V & Little I M D 1994. *India, Macroeconomics and Political Economy1964-1991* (Washinton D. C.: The World Bank).
- Marques, J., McNeilly, C. 1988. *Income and price elasticities for exports of developing countries*, Review of Economics and Statistics, Vol. 70 No.2, pp.306-14.
- Marquez, J. 2000. *The Puzzling Income Elasticity of US Imports,* Econometric Society World Congress, No 1128.
- Meliciani, Valentina, 2002. The Globalizing Learning Economy: D. Archibugi, B. Lundvall (Eds.), Oxford University Press, Oxford, 2001, pp. 307.
- Nigel, P. and Desirée V. W. 2004. *International Production Relocation And Exports of Services*, OECD, Economic Studies No. 38, January.
- Patnaik, Ila and Peter Pauly 2001. *The Indian Foreign Exchange Market and the Equilibrium Real Exchange Rate of the Rupee*, Global Business Review, Vol. 2, No.2, 195-212
- Senhadji, A. S. and Montenegro, C. E.1999. Time Series Analysis of Export Demand Equations: A Cross-Country Analysis, A Journal of the IMF, Vol. 46, no. 3 (September/December).
- Sharma, Kishor 2000. *Export Growth in India: Has FDI played a role*, Centre discussion paper no.816, July.
- Sinha, Dipendra 2001. A Note on Trade Elasticities in Asian Countries, International Trade Journal, Vol. 15, 221-237.

- Srinivasan T N 1996. 'India's Export Performance: A Comparative Analysis'.India's Economic Reforms and Development, Essays for Manmohan Singh. (OUP).
- Stefan Papaioannou and Kei-Mu Yi. 2001 *The Effects of a Booming Economy on the U.S. Trade Deficit,* Federal Reserve Bank of New York, Current Issues in Economics and Finance, Vol. 7, No.2, February.
- TD Economics 2007. American Exports: The Quiet Revolution, Bank Financial Group, November.
- Wijeweera, A., Nur, M., & Dollery, B., 2008. Bilateral Import Demand Elaticities the case of Bangladesh", International Research Journal of Finance and Economics, Issue no. 19.

## **ANNEX I: Results of Stationarity Tests**

#### 1. Stationarity Test for Logexports deflated by Export Unit Value Index

Z(t)	-0. 145	-4. 260	-3. 548	-3. 209
	Test Stati sti c	Inter 1% Critical Value	polated Dickey-Full 5% Critical Value	er 10% Critical Value
Dickey-Fuller t	est for unit ro	oot	Number of obs	= 38
. dfuller Inrea	lexpgs, trend l	ags(0)		

MacKinnon approximate p-value for Z(t) = 0.9924

. dfuller Inexpaseanuvdef, trend Iags(0)

#### 2. Stationarity Test for Exports to ASEAN5.

Z(t)	-1. 291	-4. 362	-3. 592		-3. 235
	Test Stati sti c	Inter 1% Critical Value	rpolated Dickey-Full 5% Critical Value	er – 10%	Critical Value
Di ckey-Ful I	er test for unit r	oot	Number of obs	=	27
. dfuller I	nagdpcon, trend la	gs(0)			
MacKinnon a	pproximate p-value	for Z(t) = <b>0.890</b>	l		
Z(t)	-1. 291	-4. 362	-3. 592		-3. 235
	Test Stati sti c	Inter 1% Critical Value	rpolated Dickey-Full 5% Critical Value	er — 10%	Critical Value
Di ckey-Ful I	er test for unit r	oot	Number of obs	=	27
. dfuller l	nagdpcon, trend la	gs(0)			
MacKinnon a	pproximate p-value	for Z(t) = 0.5047	7		
Z(t)	-2. 174	-4. 362	-3. 592		-3. 235
	Test Stati sti c	Inter 1% Critical Value	rpolated Dickey-Full 5% Critical Value	er – 10%	Critical Value
Dickey-Fuller test for unit root			Number of obs	=	27

MacKinnon approximate p-value for Z(t) = 0.8901

.

## 3. Stationarity Test for Exports to G7

	meerder, rags(0)				
dfuller I	nreerdef Lags(0)				
MacKi nnon a	approximate p-value	for	Z(t) = 1.000	00	· · · · · · · · · · · · ·
Z(t)	0. 915		-4. 362	-3. 592	-3. 235
	Test Stati sti c	1%	Critical Value	erpolated Dickey-Fuller 5% Critical 10 Value	Critical Value
Di ckey-Ful I	er test for unit r	oot		Number of obs =	27
. dfuller l	nreerdef, trend la	ngs(0)	)		
MacKinnon a	approximate p-value	e for	Z(t) = 0.729	91	
Z(t)	-1.748		-4. 362	-3. 592	-3. 235
	Test Stati sti c	1%	Critical Value	erpolated Dickey-Fuller 5% Critical 10 Value	% Critical Value
Di ckey-Ful I	er test for unit r	oot		Number of obs =	27
. dfuller I	ngdpg7, trend lags	6(0)			
MacKinnon a	approximate p-value	e for	Z(t) = <b>0.504</b>	17	
Z(t)	-2. 174		-4. 362	-3. 592	-3. 235
	Test Stati sti c	1%	Critical Value	5% Critical 10 Value	% Critical Value
			Inte	erpolated Dickey-Fuller	
Di ckey-Ful I	er test for unit r	oot		Number of obs =	27

.

### 4. Stationarity Tests for Sectors

Z(t)	-3. 332	-4. 270	-3. 552	-3. 211
	Test Stati sti c	1% Critical Value	polated Dickey-Full 5% Critical Value	er — 10% Critical Value
Di ckey-Ful	ler test for unit ro	ot	Number of obs	= 37
. dfuller	Inmarinedefuv, trend	lags(0)		
MacKi nnon	approximate p-value	for Z(t) = <b>0.7503</b>	3	
Z(t)	-1. 701	-4. 270	-3. 552	-3. 211
	Test Stati sti c	1% Critical Value	polated Dickey-Full 5% Critical Value	er ——— 10% Critical Value
Di ckey-Ful	ler test for unit ro	ot	Number of obs	= 37
. dfuller	Inplantdefuv, trend	l ags (0)		
MacKi nnon	approximate p-value	for Z(t) = <b>0.0489</b>	)	
Z(t)	-3. 419	-4. 270	-3. 552	-3. 211
	Test Stati sti c	Inter 1% Critical Value	polated Dickey-Full 5% Critical Value	er — 10% Critical Value
Di ckey-Ful	ler test for unit ro	ot	Number of obs	= 37
. dfuller	LNTEXTDEFUV, trend I	ags(0)		

MacKinnon approximate p-value for Z(t) = 0.0612

Z(t)	-2. 150	-4.27	0 -3.5	552	-:	3. 211
	Statistic	Val ue	Valu	e	Va	ıl ue
	Test	1% Critica	I 5% Criti	cal 1	10% Cri	tical
			Interpolated Did	ckey-Fulle	er	
Dickey-Fuller	test for unit	root	Number	of obs	=	37

MacKinnon approximate p-value for Z(t) = 0.5178

Di ckey-Ful I	ler test for unit	root	Number of obs	= 37
	Test Stati sti c	Inter 1% Critical Value	polated Dickey-Full 5% Critical Value	er ——— 10% Critical Value
Z(t)	-2. 216	-4. 270	-3. 552	-3. 211
MacKinnon a	approximate p-valu	e for Z(t) = <b>0.4806</b>	1	<u> </u>
. dfuller l	lnengdefuv, trend	ags(0)		
Di ckey-Ful I	ler test for unit	root	Number of obs	= 37
	Test Stati sti c	Inter 1% Critical Value	polated Dickey-Full 5% Critical Value	er — 10% Critical Value
Z(t)	-0. 374	-4. 270	-3. 552	-3. 211
MacKinnon a	approximate p-valu	e for Z(t) = <b>0.9876</b>		
. dfuller l	Inagri, trend lags	(0)		

Z(t)	-1. 470	-4. 270	-3. 552	-3. 211
	Test Stati sti c	Inter 1% Critical Value	rpolated Dickey-Fu 5% Critical Value	ller — 10% Critical Value
Di ckey-Ful	ler test for unit	root	Number of obs	s = <b>37</b>

MacKinnon approximate p-value for Z(t) = 0.8392

## 5. Stationarity Tests for REER

Z(t)	-2. 660	-4. 260	-3. 548	-3. 209
	Test Statistic	1% Critical Value	erpolated Dickey-Fu 5% Critical Value	Iler — 10% Critical Value
Dickey-Fuller test for unit root			Number of ob	s = <b>38</b>

MacKinnon approximate p-value for Z(t) = 0.2532

#### 6. Stationarity Tests for Log Global GDP

	dfuller	l nggdp,	trend	lags	5(0)	
Di	ckey-Ful	ler test	t for u	uni t	root	

	Test Stati sti c	1% Critical Value	erpolated Dickey-Fu 5% Critical Value	uller ——— 10% Critical Value
Z(t)	-3. 153	-4. 260	-3. 548	-3. 209

Number of obs =

38

MacKinnon approximate p-value for Z(t) = 0.0941

## ANNEX II: New and Potential Products for Exports in Developing Countries

S. no	HS Code	Potential	New
1	10420		Australia,Brazil,Korea,Russia,,,,,,
2	10612		Australia
3	10631		Australia
4	20210		Australia,Brazil,Korea,Russia,
5	20220		Australia,,,Russia,,,,,,
6	20230	UAE	Australia,,,Russia,,Malaysia,Philippines,,,South Africa,
7	20421		Australia,Brazil,Korea,Russia,,,,,,
8	20442		Australia,,,Russia,,,,,,
9	20690		Australia,,,Russia,,,,,,
10	20810		Australia
11	20830		Australia
12	20840		Australia
13	21091		Australia,Brazil,Korea,Russia,,,,,,
14	21092		Australia
15	21093		Australia
16	30232		Australia,Brazil,,Russia,,,,,South Africa,
17	30233		Australia,Brazil,Korea,Russia,,,,,,
18	30236		Australia
19	30239		Australia,Brazil,Korea,Russia,,,,,,
20	30342		Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
21	30349		,Brazil,Korea,Russia,,,,,,South Africa,
22	30379	UAE	Australia,,Korea,Russia,,Malaysia,Philippines,,,South Africa,
23	30559		""Russia,"""South Africa,
24	30569	UAE	Australia,Brazil,Korea,Russia,,,,,South Africa,
25	30611		Australia,Brazil,,Russia,,,,,,
26	30612		Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
27	30613	Australia, UAE, South Africa	,Brazil,Korea,Russia,,Malaysia,Philippines,,,,
28	30614		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
29	30624		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
30	30739		Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
31	30741	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, "South Africa,
32	30749	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
33	40210	UAE	Australia,,,Russia,,Malaysia,Philippines,,,South Africa,
34	40229		Australia, Brazil, Russia, Malaysia, Philippines, ,,,
35	40700		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,
36	40811		Australia,,Korea,Russia,UAE,,,Singapore,,South Africa,
37	40891		Australia, Brazil, Korea, Russia, UAE,,,,Singapore,,South Africa,
38	40900	UAE	,,,Russia,,Malaysia,Philippines,,,South Africa,
39	50100		Australia, Brazil, Korea, Russia, ,,, Singapore, ,South Africa,
40	50610		Australia, Brazil, Korea, Russia,,,,,,
41	50690		Australia,,,Russia,,,,,South Africa,

## Table: New and Potential Products for Exports: Six-Digit Level Analysis

42	50790		Australia,Brazil,Korea,Russia,,,,,,
43	51000		Australia,,,,Russia,,,,,,,
44	60390	South Africa	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,,
45	60499	South Africa	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,,
46	70310	UAE, Malaysia	Australia, Brazil, Korea, Russia, ,, Philippines, ,, South Africa,
47	70990	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
48	71010		Australia,Brazil,Korea,Russia,,,,,,
49	71140		Australia,Brazil,Korea,Russia,,,,,,
50	71220		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
51	71231		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
52	71320		,Brazil,,Russia,,,,,,
53	71340		Australia,Brazil,Korea,Russia,,,,,,
54	71390	Australia, UAE	,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
55	80119		Australia,Brazil,Korea,Russia,,,,,,
56	80131		Australia,Brazil,Korea,Russia,,,,,,
57	80132		Australia,Brazil,Korea,Russia,,,,,,
58	80450	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, South Africa,
59	80590		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,,South Africa,
60	80720		Australia,Brazil,Korea,Russia,
61	81090	UAE	Australia, Brazil, Korea, Russia, Malavsia, Philippines, South Africa,
62	81290		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,
63	90111	UAE	Australia, Brazil, Korea, Russia, Malavsia, Philippines, South Africa,
64	90230	UAE, South Africa	Australia, Brazil, Korea, Russia, Malavsia, Philippines,
65	90240	Brazil, UAE, Malaysia, South Africa	Australia, Korea, Russia, Philippines,
66	90411	Brazil	Australia. Korea. Russia. UAE. Malaysia. Philippines South Africa.
67	90412	South Africa	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines
68	90420	UAE	Australia, Brazil, Korea, Russia, Malavsia, Philippines, South Africa,
69	90500		Australia,Brazil,Korea,Russia,
70	90620		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,
71	90700		Brazil,Korea,Russia,,
72	90810		Australia, Brazil, Korea, Russia,
73	90820		,Brazil,Korea,Russia,,
74	90830		Australia, Brazil, Korea, Russia,
75	90910		Australia, Brazil, Korea, Russia
76	90920	Australia	,Brazil,Korea,Russia,,
77	90930		Australia,Brazil,Korea,Russia
78	90950		Australia, Brazil, Korea, Russia
79	91030		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
80	91091	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, South Africa,
81	91099	UAE, Malaysia	Australia, Brazil, Korea, Russia, "Philippines, "South Africa,
82	100590		Thailand
83	100630	UAE, Singapore, Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,South Africa,
84	100640		Australia,Brazil,Korea,Russia,,,,,,
85	100820		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
86	110220		Thailand
87	110290	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
88	110610	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
89	110630		Africa, Australia Brazil Korea Russia HAF Malaysia Philippines Singapore Thailand S
57	110050		rastana, stazi, reisa, reissia, er is, manaysia, i imppines, singapore, i nanand, s

			outh Africa,
90	120210		Australia,Brazil,Korea,Russia,,,,,,
91	120220	UAE, Malaysia	Australia,,Korea,Russia,,,Philippines,Singapore,Thailand,South Africa,
92	120300		Australia,Brazil,Korea,Russia,,,,,,
93	120740	Australia, Thailand	,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
94	120750		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa
95	120799		Australia,Brazil,Korea,Russia,,,,,,
96	120810		,,,,,,Thailand,,
97	120999		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, South Africa,
98	121130		Australia,,,,,,,
99	121190	Australia, UAE	,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
100	130190	UAE, Malaysia, Singapore	Australia, Brazil, Korea, Russia,,,,,, South Africa,
101	130211		Australia,Brazil,Korea,Russia
102	130219	Australia, Brazil, Korea, Singapore	Russia, UAE, Malaysia, Philippines, , Thailand, South Africa,
103	130232	Singapore	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines,, Thailand, South Africa,
104	140420		Australia,Brazil,Korea,Russia,,,,,,
105	140490		Australia, Brazil, Korea, Russia, Malaysia, Philippines, , Thailand, South Africa,
106	150410		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
107	150430		Australia
108	150810		Australia,,Korea,Russia
109	151319	UAE	Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
110	151491		Australia,Brazil,Korea,Russia,,,,,,
111	151530		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
112	151550	Singapore	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, , Thailand, South Africa,
113	151590		Thailand
114	151620	UAE	,Brazil,Korea,Russia,,,,,South Africa,
115	151800	Malaysia	"Korea, Russia, "Philippines, Singapore, Thailand, South Africa,
116	160420		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,
117	160520		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
118	160530		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
119	170111		Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,,
120	170191		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
121	170199	Malaysia, Singapore	Australia, Brazil, Korea, Russia, ,, Philippines, , Thailand, South Africa,
122	170230		Thailand
123	170310		Australia,Brazil,Korea,Russia,,,,,,
124	190300		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
125	190490	Malaysia, Singapore	Australia,Brazil,Korea,Russia,UAE,,Philippines,,,South Africa,
126	200110	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
127	200190	UAE, Singapore, Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
128	200510	Singapore	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,
129	200799	UAE, Malaysia, Singapore	Australia,,Korea,Russia,,,Philippines,,Thailand,South Africa,
130	210111	UAE, Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
131	210120	Malaysia, Singapore, Thailand	Australia,Brazil,Korea,Russia,UAE,,Philippines,,,South Africa,

132	210130		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
133	220720		Thailand
134	230400		Australia,,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
135	230500		Australia,Brazil,Korea,Russia,,,,,,
136	230630		Australia,,Korea,Russia,,,,,,
137	230649		Australia,Brazil,Korea,Russia,,,,,,
138	230690		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, S
120	240110		outh Africa,
139	240110	Philipping	Australia, Diazii, Korea, Russia, Malaveia South Africa
140	240120	rimppines	Australia, Korea Dussia UAE Malaysia Dhilinning Singapore Thailand South
141	240130		Africa,
142	240399	UAE, Malaysia	Australia, Brazil, Korea, Russia, ,, Philippines, Singapore, Thailand, South Africa,
143	250100	UAE, Singapore, Thailand, South Africa	Australia,Brazil,,Russia,,Malaysia,Philippines,,,,
144	250510		Thailand
145	250610		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
146	250629		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
147	250810	Australia, Singapore, Thailand	"Korea, Russia, UAE, Malaysia, Philippines, "South Africa,
148	251110		Australia,Brazil,Korea,Russia,,,,,,,
149	251320	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
150	251400		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa.
151	251512	Australia	,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
152	251611		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand,
153	251612	South Africa	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand,,
154	251622		Australia,Brazil,Korea,Russia,,,,,,
155	251690		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
156	251830		Australia, Brazil, Korea, Russia,,,,,,
157	252100		Australia, Brazil, Korea, Russia, ,, Philippines, Singapore, Thailand, South Africa,
158	252321	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
159	252329	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
160	252510		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa.
161	252520		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa.
162	252530		Australia,Brazil,Korea,Russia,,,,,,
163	252910	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
164	260111		Australia, Brazil, Russia,,,,Singapore,,South Africa,
165	260112		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
166	260200		Australia, Brazil, Russia, ,,, Singapore, South Africa,
167	260600	UAE	Australia, Brazil, Korea, Russia,,,,Singapore,,South Africa,
168	260700		Australia, Brazil, Russia, ,,, Singapore, South Africa,
169	260800		Australia,Brazil,,Russia,,,,,,
170	261000		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
171	261210		Australia,,,,,,,
172	261400	Malaysia	Australia, Brazil, Russia, ,, Philippines, Singapore, Thailand, South Africa,
173	261690		Australia,Brazil,Korea,Russia,,,,,,

174	261800		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
175	261900		Atrica, Australia, Brazil, Korea, Russia, Malaysia, Philippines, South Africa,
176	262030		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
177	262110		Australia,Brazil,Korea,Russia,,,,,,
178	262190	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
179	270119		Australia,Brazil,Korea,Russia,,,,,,
180	270500		Australia,Brazil,Korea,Russia,,,,,,
181	270710		Australia, Brazil, Korea, Russia, UAE,,, Singapore,, South Africa,
182	270740		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
183	270799		Thailand
184	271011		"Korea,Russia,,Malaysia,Philippines,,,,
185	271019	Australia, Korea, UAE, Malaysia, Philippines, South Africa	,Brazil,Russia
186	271099		Australia,Brazil,Korea,Russia,,,,,,South Africa,
187	271111		Australia,Brazil,,Russia,,,,,,
188	271210	Korea, Singapore	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
189	271312		Thailand
190	280200		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
191	280300	UAE, Malaysia	Australia,,,Russia,,,Philippines,,Thailand,South Africa,
192	280700		Australia,Brazil,Korea,,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
193	280910		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
194	281390		Australia,Brazil,Korea,Russia,UAE,,,Singapore,,South Africa,
195	281530		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
196	281700	Singapore	Australia,,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
197	281820		Australia,Brazil,,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
198	281830		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
199	282120		Australia,Brazil,Korea,Russia,,,,,,
200	282300	Korea, UAE, Singapore	Australia,Brazil,,Russia,,Malaysia,Philippines,,Thailand,South Africa,
201	282612	UAE	Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
202	282732	Australia	,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
203	282739		Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,
204	282760		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
205	282810		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
206	282919		Australia,Brazil,Korea,Russia,,,,,,
207	282990		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
208	283190		Australia, Brazil, Korea, Russia,,,,,,,
209	283220		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
210	283319		Australia,Brazil,,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
211	283321		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
212	283322		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
213	283330		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
214	283711		Thailand

215	284130		Thailand
216	284150		Thailand
217	284800		Thailand
218	290124		Australia,Brazil,Korea,Russia,UAE,,,,,South Africa,
219	290220		Australia, Brazil, Korea, UAE, Malaysia, Philippines, ,, South Africa,
220	290241		Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
221	290243		Australia, Brazil, Korea, Russia,,, Philippines, Singapore, Thailand, South Africa,
222	290244		Australia, Brazil, Korea, "Malaysia, Philippines, Singapore, Thailand, South Africa,
223	290290	Korea, UAE, Singapore	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
224	290319		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
225	290341		Australia, Brazil, Korea, Russia,
226	290342		Australia,Brazil,Korea,Russia
227	290344		Australia,Brazil,Korea,Russia
228	290345		Australia,Brazil,Korea,,,,,,South Africa,
229	290349	UAE, Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
230	290361		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
231	290362		Australia,Brazil,Korea,Russia,,,,,,
232	290369		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
233	290410	Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
234	290420		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
235	290490		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
236	290514		Australia, Brazil, Korea, Russia, ,, Philippines, ,, South Africa,
	200517	Malaysia	Australia Brazil Korea Russia South Africa
237	290517	Ivialay Sia	Australia, Diazli, Robel, Russia,,,,,,South Annea,
237 238	290517 290522	iviaiaysia	Thailand
237 238 239	290517 290522 290531		Thailand ,Brazil,Korea,Russia,,,,,,
237 238 239 240	290517 290522 290531 290611		Thailand ,Brazil,Korea,Russia,,,,,,, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
237 238 239 240 241	290517 290522 290531 290611 290619	Singapore	Thailand         ,Brazil,Korea,Russia,,,,,,,,,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South         Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
237 238 239 240 241 242	290517 290522 290531 290611 290619 290621	Singapore	Thailand         ,Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243	290517 290522 290531 290611 290619 290621 290629	Singapore Singapore Singapore	Thailand         ,Brazil,Korea,Russia,,,Malaysia,Philippines,Singapore,Thailand,South         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
237 238 239 240 241 242 243 244	290517 290522 290531 290611 290619 290621 290629 290722	Singapore Singapore Singapore	Thailand         ,Brazil,Korea,Russia,,,,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,
237 238 239 240 241 242 243 244 244 245	290517 290522 290531 290611 290619 290621 290629 290722 290722 290729	Singapore Singapore Singapore Singapore	Thestulin, Drazil, Korea, Russia, ,, Molaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, South Africa,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, South Africa,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, South Africa,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, ,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, ,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, ,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, ,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, ,         Australia, Brazil, Korea, Russia, ,, Malaysia, Philippines, ,, Thailand, South Africa,
237 238 239 240 241 242 243 244 245 246	290517 290522 290531 290611 290619 290621 290629 290722 290722 290729 290890	Singapore Singapore Singapore Singapore	Thailand         ,Brazil,Korea,Russia,,,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
237 238 239 240 241 242 243 244 245 244 245 246 247	290517 290522 290531 290611 290619 290621 290629 290722 290722 290729 290890 290911	Singapore Singapore Singapore Singapore	Thestunin, Brazil, Korea, Russia, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 244 245 246 247 248	290517 290522 290531 290611 290619 290621 290629 290722 290722 290729 290890 290911 290941	Singapore Singapore Singapore Singapore	Thailand         ,Brazil,Korea,Russia,,,,,,,,,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,         Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,         Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa,         Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa,
237 238 239 240 241 242 243 244 245 245 246 247 248 249	290517 290522 290531 290611 290619 290621 290629 290722 290729 290729 290890 290911 290941 290950	Singapore Singapore Singapore Singapore Singapore Singapore	Thailand         Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 246 247 248 249 250	290517 290522 290531 290611 290619 290621 290629 290722 290729 290729 290890 290911 290941 290950 291219	Singapore Singapore Singapore Singapore Singapore Singapore Singapore Singapore	Thailand         Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 244 245 246 247 248 249 250 251	290517 290522 290531 290611 290619 290621 290629 290722 290729 290729 290890 290911 290941 290941 290950 291219 291221	Singapore Singapore Singapore Singapore Singapore Singapore Singapore	Thailand         Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 245 246 247 248 249 250 251 252	290517 290522 290531 290611 290619 290621 290629 290722 290729 290890 290911 290941 290941 290950 291219 291221 291249	Singapore Singapore Singapore Singapore Singapore Singapore Singapore Singapore	Thailand         Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 246 247 248 249 250 251 252 253	290517 290522 290531 290611 290619 290621 290629 290722 290729 290890 290911 290941 290941 290950 291219 291221 291249 291419	Singapore	Thailand         Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 244 245 246 247 248 249 250 251 252 253 254	290517 290522 290531 290611 290619 290621 290629 290722 290729 290729 290890 290911 290941 290941 290950 291219 291221 291249 291419 291423	Singapore	Thailand         Brazil,Korea,Russia,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
237 238 239 240 241 242 243 244 245 244 245 246 247 248 249 250 251 252 253 254 255	290517 290522 290531 290611 290619 290621 290629 290722 290729 290729 290890 290911 290941 290950 291219 291221 291221 291249 291419 291423 291440	Singapore Korea, Singapore	Thestulia, Brazil, Korea, Russia, J., Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J., Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J., Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J., Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J., Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J., Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J., Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J. (AE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J. (AE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, J. (AE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, R
237 238 239 240 241 242 243 244 245 246 247 248 249 250 251 252 253 254 255 256	290517 290522 290531 290611 290619 290621 290629 290722 290729 290729 290890 290911 290941 290941 290950 291219 291221 291249 291249 291419 291423 291440 291470	Singapore	Thestulia, Brazil, Korea, Russia, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,

258	291531		"Korea,Russia,,,,Singapore,,South Africa,
259	291539	Brazil, UAE, Singapore	Australia,,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
260	291540		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa.
261	291570	Korea	Australia, Brazil, Russia, ,, Philippines, ,, South Africa,
262	291631	Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
263	291632		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
264	291634		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
265	291735		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
266	291811	Brazil	Australia,,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
267	291813		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
268	291816		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
269	291821		Australia,,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
270	291823	Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
271	292090	Korea	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
272	292111	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
273	292141		Australia,Brazil,Korea,Russia,,,,,,
274	292142	Korea	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
275	292143	Korea	Australia, Brazil, Russia, ,,, Singapore, ,South Africa,
276	292151		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
277	292159		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
278	292214		Australia,Brazil,Korea,Russia,,,,,,
279	292221		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
280	292229		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
281	292243		Australia,Brazil,Korea,Russia,,,,,,
282	292320		Australia,,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
283	292390		Thailand
284	292411	Thailand	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, South Africa,
285	292419	Brazil	Australia,,,Russia,,,,,South Africa,
286	292520	Brazil	Australia,,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
287	292630		Australia
288	292800		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
289	293291		Australia,Brazil,Korea,Russia
290	293294		Australia
291	293311		Australia,Brazil,Korea,Russia,, Singapore,,South Africa,
292	293319		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
293	293329	Singapore	Korea,Russia,,South Africa,
294	293331		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
295	293332		Thailand,,
296	293354		Australia,Brazil,Korea,Russia
297	293355		Australia,Brazil,Korea,Russia
298	293372		Australia,Brazil,Korea,Russia
299	293626		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
300	293629		,Brazil,Korea,Russia,UAE,,,,,South Africa,

301	293739		Australia,Brazil,Korea,Russia
302	293740		Thailand
303	293911		Australia,Brazil,Korea,Russia
304	293919		Australia,Brazil,Korea,Russia
305	293929		Australia,Brazil,Korea,Russia
306	293941		Australia,Brazil,Korea,Russia
307	293942		Australia,Brazil,Korea,Russia
308	293943		Australia
309	293949		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
310	293951		Australia
311	293961		Australia
312	293991		Australia,Brazil,Korea,Russia,,,,,,
313	294110	Korea Singapore	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
314	294190	Brazil, Korea, UAE, Singapore, Thailand	Australia,,,Russia,,Malaysia,Philippines,,,South Africa,
315	294200	Korea, UAE, Philippines, Singapore	Australia,Brazil,,Russia,,Malaysia,,,Thailand,South Africa,
316	300120		Australia,,Korea,Russia,,,,Singapore,,South Africa,
317	300310		Australia,Brazil,Korea,Russia,,,,,,
318	300320	Australia	Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
319	300339		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
320	300390	Australia, UAE, Philippines, Singapore, Thailand, South Africa	Brazil,Korea,Russia,,Malaysia,,,,,
321	300410		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
322	300420	Australia,Korea,Russia,UAE,Philippi nes	,Brazil,,Malaysia,,Singapore,Thailand,South Africa,
323	300450	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
324	300680		Australia,Brazil,Korea,Russia,,,,,,
225			Australia
525	310240		
323	310240 310520		Australia
323 326 327	310240 310520 310530		Australia Australia
323           326           327           328	310240           310520           310530           310551		Australia Australia Australia
323           326           327           328           329	310240           310520           310530           310551           310560		Australia Australia Australia Australia
323           326           327           328           329           330	310240           310520           310530           310551           310560           320110		Australia Australia Australia Australia Australia,,Korea,Russia
323       326       327       328       329       330       331	310240         310520         310530         310551         310560         320110         320190	Brazil	Australia Australia Australia Australia Australia,,Korea,Russia Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
323         326         327         328         329         330         331         332	310240         310520         310530         310551         310560         320110         320190         320210	Brazil Singapore, Thailand	Australia Australia Australia Australia Australia,,Korea,Russia Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
323         326         327         328         329         330         331         332         333	310240         310520         310530         310551         310560         320110         320190         320210         320412	Brazil Singapore, Thailand Malaysia, Singapore, Thailand	Australia Australia Australia Australia Australia,Korea,Russia Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa, Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
323           326           327           328           329           330           331           332           333           334	310240         310520         310530         310551         310560         320110         320190         320210         320412         320413	Brazil Singapore,Thailand Malaysia,Singapore,Thailand Korea,UAE	Australia Australia Australia Australia Australia Australia,,Korea,Russia Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Philippines,,South Africa, Australia,Brazil,Russia,Malaysia,Philippines,,Thailand,South Africa,
323           326           327           328           329           330           331           332           333           334           335	310240         310520         310530         310551         310560         320110         320190         320210         320412         320413         320414	Brazil Singapore, Thailand Malaysia, Singapore, Thailand Korea, UAE Singapore	Australia Australia Australia Australia Australia Australia,,Korea,Russia Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Philippines,,South Africa, Australia,Brazil,Russia,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
323           326           327           328           329           330           331           332           333           334           335           336	310240         310520         310530         310551         310560         320110         320190         320210         320412         320413         320414         320415	Brazil Singapore, Thailand Malaysia, Singapore, Thailand Korea, UAE Singapore	Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia,, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia,, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,
323           326           327           328           329           330           331           332           333           334           335           336           337	310240 310520 310530 310551 310560 320110 320190 320210 320412 320413 320414 320415	Brazil Singapore,Thailand Malaysia,Singapore,Thailand Korea,UAE Singapore Korea,Singapore,Thailand	Australia Australia Australia Australia Australia Australia,Korea,Russia Australia,,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Russia,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Russia,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
323           326           327           328           329           330           331           332           333           334           335           336           337           338	310240         310520         310520         310530         310551         310560         320110         320190         320210         320412         320413         320414         320415         320416         320417	Brazil Singapore, Thailand Malaysia, Singapore, Thailand Korea, UAE Singapore Korea, Singapore, Thailand Australia, Korea, UAE, Malaysia, Singa pore, Thailand	Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,
323           326           327           328           329           330           331           332           333           334           335           336           337           338           339	310240 310520 310520 310551 310560 320110 320190 320210 320412 320413 320414 320415 320416 320417 320419	Brazil Singapore,Thailand Malaysia,Singapore,Thailand Korea,UAE Singapore Korea,Singapore,Thailand Australia,Korea,UAE,Malaysia,Singa pore,Thailand Korea,UAE, Singapore	Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
323           326           327           328           329           330           331           332           333           334           335           336           337           338           339           340	310240         310520         310520         310530         310551         310560         320110         320190         320210         320412         320413         320414         320415         320416         320417         320419         320420	Brazil Brazil Singapore, Thailand Malaysia, Singapore, Thailand Korea, UAE Singapore Korea, Singapore, Thailand Australia, Korea, UAE, Malaysia, Singa pore, Thailand Korea, UAE, Singapore Korea, Singapore	Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
323           326           327           328           329           330           331           332           333           334           335           336           337           338           339           340           341	310240 310520 310550 310551 310560 320110 320190 320210 320210 320412 320413 320414 320414 320415 320416 320417 320419 320420 320490	Brazil Singapore, Thailand Malaysia, Singapore, Thailand Korea, UAE Singapore Korea, Singapore, Thailand Australia, Korea, UAE, Malaysia, Singa pore, Thailand Korea, UAE, Singapore Korea, Singapore	Australia         Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, Thailan
323           326           327           328           329           330           331           332           333           334           335           336           337           338           339           340           341	310240 310520 310520 310551 310560 320110 320190 320210 320412 320413 320414 320415 320416 320416 320417 320419 320420 320420 320500	Brazil Singapore,Thailand Malaysia,Singapore,Thailand Korea,UAE Singapore Korea,Singapore,Thailand Australia,Korea,UAE,Malaysia,Singa pore,Thailand Korea,UAE, Singapore Korea,Singapore Korea,Singapore	Australia         Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
323           326           327           328           329           330           331           332           333           334           335           336           337           338           339           340           341           342	310240 310520 310520 310551 310560 320110 320190 320210 320412 320413 320414 320415 320416 320416 320417 320419 320420 320420 320490 320500 320620	Brazil Singapore,Thailand Malaysia,Singapore,Thailand Korea,UAE Singapore Korea,Singapore,Thailand Australia,Korea,UAE,Malaysia,Singa pore,Thailand Korea,UAE, Singapore Korea,Singapore Korea,Singapore Korea,Singapore UAE	Australia         Australia         Australia         Australia         Australia, Korea, Russia         Australia, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,         Australia, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Thailand, South Africa,         Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, UAE, Malaysia, Philippines, South Africa,         Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,         Australia, Br

	 I		Africa,
345	321519	Australia,UAE,Malaysia,Thailand	,Brazil,,Russia,,,Philippines,,,,
346	330119	Malaysia,Singapore	Australia,Brazil,Korea,Russia,UAE,,Philippines,,Thailand,South Africa,
347	330124		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
348	330125		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa.
349	330129	Australia,Brazil,UAE	""Russia, Malaysia, Philippines, Thailand,
350	330130		,,,,,,Thailand,,
351	330190	Australia,Brazil,Korea,UAE,Singapor e	,,,Russia,,Malaysia,Philippines,,Thailand,South Africa,
352	330610	UAE,Malaysia,Singapore	Australia,Brazil,Korea,Russia,,,Philippines,,Thailand,South Africa,
353	330741	UAE,Malaysia,Singapore,Thailand	Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
354	340510	Singapore	Australia,,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
355	350110		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
356	350219		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
357	350300	Singapore	Australia,,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
358	360300		Thailand,,
359	360500		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
360	370610	Australia, Philippines	Korea,Russia,UAE,Malaysia,,Singapore,,South Africa,
361	370690		Australia,Brazil,Korea,Russia,,,,,,
362	380810	UAE,Malaysia,Philippines,Singapore	Australia,Brazil,Korea,Russia,,,,,Thailand,South Africa,
363	380890	Malaysia, Philippines, Singapore, Thail and	Australia,Brazil,Korea,Russia,UAE,,,,,South Africa
364	381220		Thailand,,
365	381230	Korea, Russia, Malaysia, Philippines	Australia,Brazil,,,UAE
366	381511		Thailand,,
367	381700		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
368	382311	South Africa	Australia,,,Russia,,Malaysia,Philippines,,,,
369	382370	Malaysia	Australia,Brazil,Korea,Russia,,,,,,
370	390120		"Korea,,,,,,,
371	390210	Brazil,UAE,Malaysia	,,,,,Philippines,,,South Africa,
372	390319		,Brazil,Korea,,UAE,,,,,South Africa,
373	390760	UAE	Australia,Brazil,,,,Malaysia,Philippines,Singapore,,South Africa,
374	391212		,,,,,,,Thailand,,
375	391220		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
376	391731	Malaysia,Singapore,Thailand	Australia,Brazil,,Russia,,,Philippines,,,,
377	392069	Australia,Korea,UAE,Singapore	,Brazil,,Russia,,Malaysia,Philippines,,Thailand,South Africa,
378	392071		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
379	392329	Australia,UAE,Malaysia	,Brazil,,Russia,,,Philippines,,,,
380	400110		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
381	400121	Korea	,Brazil,,Russia,UAE,,,Singapore,,South Africa,
382	400122		,Brazil,Korea,Russia,,,,,,
383	400129		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
384	400280		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
385	400300	Malaysia	Australia,Brazil,Korea,Russia,UAE,,Philippines,Singapore,Thailand,South Africa,
386	400400		,,,,,,,Thailand,,
387	400700	South Africa	Australia,Brazil,Korea,Russia,UAE,,Philippines,,,,

388	400821	Australia,UAE,Malaysia,Thailand	,Brazil,Korea,Russia,,,Philippines,Singapore,,South Africa,
389	400921	Russia,UAE	Australia,Brazil,Korea,,,Malaysia,Philippines,,,South Africa,
390	401011	Singapore, Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
391	401012	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South A frica
392	401120	Brazil,Korea,UAE,Malaysia,Philippin es Singapore South Africa	Australia,,,Russia,,,,,,
393	401150	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
394	401192		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
395	401199	UAE,Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
396	401290	Brazil,Malaysia,Singapore,Thailand,S outh Africa	Australia,,Korea,Russia,UAE,,Philippines,,,,
397	401320		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
398	401390	Singapore	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
399	401410		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
400	401511	Malaysia	Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
401	401691	South Africa	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,,
402	401695		Australia,,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
403	401700	Australia,UAE	,Brazil,Korea,Russia,,Malaysia,,Singapore,,South Africa,
404	410419	Russia	Australia,,Korea,,Malaysia,,,South Africa,
405	410449	Korea, Malaysia	Australia,,,Russia,,,Philippines,,,South Africa,
406	410510		Australia,Brazil,Korea,Russia,,Singapore
407	410530		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa
408	410621		Australia,Brazil,Korea,Russia,,,,,,
409	410622		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
410	410640		Australia, Korea, Russia,
411	410691		Australia,Brazil,Korea,Russia,
412	410692		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
413	410711	Korea	Brazil, Russia, Malaysia, Philippines, South Africa,
414	410719	Korea	Australia,,,Russia,UAE,Malaysia,,,,South Africa,
415	410799	Korea, Malaysia, South Africa	Australia,Brazil.,Russia,Philippines,
416	411200	Korea,UAE	Australia, Brazil, Russia, Malaysia, Philippines, South Africa,
417	411310	Korea.UAE	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
418	411390	,	Australia Brazil. Russia. Malavsia Philippines
419	420100	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South
420	420221	UAE,Malaysia,Singapore,Thailand	Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
421	420231	Australia,UAE,Malaysia,Singapore	,Brazil,Korea,Russia,,,Philippines,,Thailand,South Africa,
422	420310	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
423	420329	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
424	420340	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
425	420610		Australia,Brazil,Korea,Russia,,,,,,
426	440420		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
427	440810		Thailand
428	440831		Australia,Brazil,Korea,Russia,,,,,,
429	480210		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa
430	480220	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
431	480254		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,

432	480269	UAE	Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
433	481039		Thailand
434	481099	Korea,UAE	Australia,Brazil,,Russia,,Malaysia,Philippines,,Thailand,South Africa,
435	482020		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa
436	482390	Australia,Korea,Russia,UAE,Malaysi a,Philippines,Singapore,Thailand,Sou th Africa	Brazil
437	490110	Australia,Korea,UAE,Malaysia,Thail and,South Africa	,Brazil,Philippines,
438	500100		Australia,Brazil,Korea,Russia,,,,,,
439	500600		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
440	500710		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
441	500720	Australia,Korea,Thailand	,Brazil,,Russia,,Malaysia,Philippines,Singapore,,South Africa,
442	500790	Korea,UAE,Singapore,Thailand	Australia, Brazil, Russia, Malaysia, Philippines, ,, South Africa,
443	510400		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
444	511111		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
445	511120		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
446	511190		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
447	520100		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
448	520210		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
449	520299		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
450	520300		,,,,,,Thailand,,
451	520411		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
452	520419		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
453	520420	Malaysia	Australia,Brazil,Korea,Russia,,,Philippines,Singapore,Thailand,South Africa,
454	520511		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
455	520512		,,,,,,Thailand,,
456	520513		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
457	520515		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
458	520521		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
459	520522		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
460	520523		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
461	520524		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
462	520526		Australia, Brazil, Korea, Russia,,,,,Singapore,,,South Africa,
463	520527		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
464	520528		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
465	520533		Australia, Brazil, Korea, Russia, ,,,, Singapore,,, South Africa,
466	520535		Australia, Brazil, Korea, Russia, ,,,, Singapore,,, South Africa,
467	520541		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
468	520546		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
469	520547		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
470	520548		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
471	520611		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
472	520621		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South

			Africa,
473	520633		Australia,,,,,,
474	520634		Australia,,,,,,
475	520710		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
476	520790	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
477	520811		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
478	520812	Malaysia	Australia, Brazil, Korea, Russia, ,, Philippines, Singapore, Thailand, South Africa,
479	520821	Korea, Thailand	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, South Africa,
480	520823		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
481	520829		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
482	520831	Korea	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, South Africa,
483	520841		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
484	520851	Korea,Singapore	Australia, Brazil, Russia, Malaysia, Philippines, ,, South Africa,
485	520859	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
486	520911		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
487	520912	Malaysia	Australia, Brazil, Korea, Russia, "Philippines, Singapore, Thailand, South Africa,
488	520921		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
489	520922		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
490	520931		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
491	520941		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
492	520942	Malaysia, Thailand	Australia, Brazil, Korea, Russia, "Philippines, Singapore, South Africa,
493	520951		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
494	521011		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
495	521019		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
496	521029		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
497	521051		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
498	521111		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
499	521131		Australia,Brazil,,,,Malaysia,Philippines,Singapore,Thailand,South Africa,
500	521142		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
501	521151		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
502	521211		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
503	521212	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
504	521213	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
505	521214		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
506	521215	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
507	521222		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
508	521223		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
509	521225		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
510	530121		Australia,,,,,,,,
511	530310		Australia,Brazil,Korea,Russia,,,,,,

512	530390		Australia,Brazil,Korea,Russia,,,,,,
513	530620		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
514	530710		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
515	530720		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
516	530810		Australia,Brazil,Korea,Russia,,,,,,
517	531010		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
518	531090		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
519	540233	Korea,UAE,Malaysia	Australia, Brazil, Russia, Philippines, South Africa,
520	540242		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
521	540251		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
522	540252		Australia, Brazil, Korea, "Malaysia, Philippines, "South Africa,
523	540262	Thailand,South Africa	Australia, Brazil, Russia, UAE, Malaysia, Philippines, Singapore,,,
524	540310		Australia,Brazil,Korea,Russia,,,,,,
525	540331	Brazil	Australia,,Korea,Russia,,,,Singapore,,South Africa,
526	540342		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
527	540610	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
528	540710	Brazil,Korea,UAE,Thailand	Australia,,,Russia,,Malaysia,Philippines,Singapore,,South Africa,
529	540720	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
530	540730		,,,,,,Thailand,,
531	540742		Australia, Brazil, Korea, "Malaysia, Philippines, Singapore, "South Africa,
532	540744		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
533	540774		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
534	540781		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
535	540784		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
536	540794		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
537	540810		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
538	540822		Australia,Brazil,,,,Malaysia,Philippines,Singapore,Thailand,South Africa,
539	540824		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
540	540831		Australia, Brazil, Korea, "Malaysia, Philippines, Singapore, Thailand, South Africa,
541	540834	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
542	550320	Singapore	Australia,Brazil,,,UAE,,Philippines,,,South Africa,
543	550410		Australıa,Brazil,Korea,Russia,,,,Singapore,,South Africa,
544	550510		Australia,Brazil,Korea,Russia,UAE,,,Singapore,,South Africa,
545	550921		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
546	550922	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
547	550931		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
548	550941		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
549	550942		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
550	550951	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
551	550952		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
552	550953		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,

553	550959		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, S
554	550000		outh Africa, Australia Brazil Korea Russia, Malaysia Philippines Singapore Thailand South
554	550777		Africa,
555	551011		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
556	551012		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
557	551030		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
558	551130		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, S outh Africa,
559	551219	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
560	551221		Australia, Brazil, Korea, "Malaysia, Philippines, Singapore, Thailand, South Africa,
561	551229		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
562	551329		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, S outh Africa,
563	551349	UAE,Singapore,Thailand	Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
564	551449		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
565	551511	Thailand	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, South Africa,
566	551512		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
567	551513	Korea,Singapore	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
568	551519	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, South Africa,
569	551591		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
570	551623		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
571	551634		Australia,,,,,,,,
572	551642		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
573	551644		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
574	560121	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
575	560129		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand,,
576	560221		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
577	560729		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
578	560749	Singapore	Australia, Brazil, Korea, "Malaysia, Philippines, "South Africa,
579	560750	Russia,Singapore,South Africa	Australia,Brazil,,,,Malaysia,Philippines,,,,
580	560790	Singapore, Thailand	Australia, Brazil, Korea, Russia, , Malaysia, ,,, South Africa,
581	560890	Malaysia	Australia,Brazil,Korea,Russia,UAE,,Philippines,Singapore,Thailand,South Africa,
582	560900	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
583	570110		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
584	570190	UAE, Philippines, Singapore	Australia, Brazil, Korea, Russia, "Malaysia, "Thailand, South Africa,
585	570210		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
586	570220		Australia,Brazil,Korea,Russia,,,,,,
587	570231	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
588	570232		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore,,,
589	570239		Australia,Brazil,Korea,Russia,,,,,,
590	570241	Malaysia	Australia, Brazil, Korea, Russia, "Philippines, Singapore, Thailand,
591	570249		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,

592	570259		Australia,Brazil,Korea,Russia,,,,,,
593	570291		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa.
594	570299	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa
595	570310	Malaysia	Australia,Brazil,Korea,Russia,,,Philippines,Singapore,,South Africa,
596	570390	South Africa	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,,
597	570500	Australia,UAE,Malaysia,Singapore,T hailand South Africa	,Brazil,Korea,Russia,,,Philippines,,,,
598	580190		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa
599	580211		Australia,,Korea,Russia,,,,,,
600	580219		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa
601	580220		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
602	580430		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
603	580500		Thailand,,
604	580810		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
605	580890	UAE, Thailand, South Africa	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore,,,
606	580900		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
607	581099	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, South Africa,
608	581100	Thailand	
609	590290		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
610	590699	UAE,Malaysia	Australia, Brazil, Korea, Russia, ,, Philippines, , Thailand, South Africa,
611	600121		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
612	600129		Thailand,,
612 613	600129 600310		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612 613 614	600129           600310           600320		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612 613 614 615	600129           600310           600320           600390		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
612 613 614 615 616	600129           600310           600320           600390           600521		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa, Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
612 613 614 615 616 617	600129           600310           600320           600390           600521           600522		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,,Singapore,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612         613         614         615         616         617         618	600129           600310           600320           600390           600521           600522           600524		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619	600129           600310           600320           600390           600521           600522           600524           610190		Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619           620	600129           600310           600320           600390           600521           600522           600524           610190           610290	UAE	Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621	600129           600310           600320           600390           600521           600522           600524           610190           610319	UAE	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621           622	600129           600310           600320           600390           600521           600522           600524           610190           610319           610413	UAE	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621           622           623	600129           600310           600320           600390           600521           600522           600524           610190           610319           610413           610419	UAE	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,,,,,,, Australia,Brazil,Korea,Russia,,,,,,,, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621           622           623           624	600129           600310           600320           600320           600521           600522           600524           610190           610290           610413           610442	UAE Singapore	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Walaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Walaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Walaysia,Philippines,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621           622           623           624           625	600129           600310           600320           600320           600320           600521           600522           600524           610190           610290           610319           610413           610442           610510	UAE Singapore Korea,Malaysia,Philippines,Singapor e	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Rorea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Russia,,,,,South Africa,
612           613           614           615           616           617           618           619           620           621           622           623           624           625           626	600129           600310           600320           600320           600390           600521           600522           600524           610190           610290           610413           610413           610410           610510           610590	UAE Singapore Korea,Malaysia,Philippines,Singapor UAE,Singapore	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621           622           623           624           625           626           627	600129           600310           600320           600320           600521           600522           600524           610190           610290           610413           610419           610510           610590           610610	UAE UAE Singapore Korea,Malaysia,Philippines,Singapor e UAE,Singapore Korea,Thailand	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,South Africa,
612         613         614         615         616         617         618         619         620         621         622         623         624         625         626         627         628	600129           600310           600320           600320           600320           600521           600522           600524           610190           610290           610319           610413           610442           610510           610590           610610           610690	UAE Singapore Korea,Malaysia,Philippines,Singapor e UAE,Singapore Korea,Thailand	Thailand,, Australia,Brazil,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,
612         613         614         615         616         617         618         619         620         621         622         623         624         625         626         627         628         629	600129           600310           600320           600320           600390           600521           600522           600524           610190           610290           610319           610413           610442           610510           610590           610610           610690           610711	UAE Singapore Korea,Malaysia,Philippines,Singapor e UAE,Singapore Korea,Thailand UAE	Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
612           613           614           615           616           617           618           619           620           621           622           623           624           625           626           627           628           629           630	600129           600310           600320           600320           600320           600521           600522           600524           610190           610290           610319           610413           610419           610442           610510           610590           610610           610690           610711           610719	UAE Singapore Korea,Malaysia,Philippines,Singapor e UAE,Singapore Korea,Thailand UAE	Thailand,, Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa, Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,

			Africa,
632	610729		Australia,Brazil,Korea,Russia,,,,,,
633	610811		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
634	610819		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa
635	610829		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
636	610831		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa.
637	610839		Australia, Brazil, Korea, Russia,,,,,,
638	610910	Australia,Korea,UAE,Malaysia,Singa pore.Thailand	Russia,,,Philippines,,,South Africa,
639	611011	<b>F j</b> =	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
640	611120		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
641	611190		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,
642	611420	Australia,UAE,Singapore	,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
643	611520		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
644	611599	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
645	620319	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
646	620342	Korea,UAE,Philippines,Singapore	Australia,Brazil,,Russia,,Malaysia,,,,South Africa,
647	620411		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
648	620412		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
649	620413		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
650	620419	South Africa	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand,
651	620421		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
652	620422		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
653	620441		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
654	620442	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
655	620443	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
656	620449	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
657	620452	Philippines,Singapore,Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,,,,South Africa,
658	620453	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
659	620461		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
660	620520	Australia,Korea,UAE,Malaysia,Singa pore	,Brazil,,Russia,,,Philippines,,,South Africa,
661	620590	UAE,Singapore,South Africa	Australia,Brazil,Korea,Russia,,Malaysia,Philippines
662	620610	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
663	620620		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
664	620630	UAE,Singapore,Thailand	Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
665	620640	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
666	620690	UAE, Thailand, South Africa	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore,,,
667	620791	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
668	620799		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
669	620811		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
670	620819		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, Singapore, The State St

			outh Africa,
671	620821		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa
672	620829	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
673	620891		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
674	620892		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa.
675	620920	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
676	620990		Australia, Brazil, Korea, Russia,,,,,,
677	621112		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
678	621410	Korea,Malaysia	Australia, Brazil, Russia, ,, Philippines, Singapore, Thailand, South Africa,
679	621420		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
680	621430	Korea	Australia,Brazil,,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
681	621440		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
682	621490	UAE,Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
683	621590	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
684	630120		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
685	630130	UAE,South Africa	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand,,
686	630190	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, , Thailand, South Africa,
687	630210		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
688	630221	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
689	630222	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
690	630229		Australia,Brazil,Korea,Russia,,,,,,
691	630239		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
692	630240		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
693	630259		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
694	630319	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
695	630391	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
696	630399	UAE,Malaysia	Australia, Brazil, Korea, Russia,,, Philippines, Singapore, Thailand, South Africa,
697	630419	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
698	630492	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
699	630499	UAE,Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
700	630510		Australia,Brazil,Korea,Russia,,,,,,
701	630520		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
702	630532		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
703	630710	Korea, Russia, Singapore, Thailand, Sou th Africa	Australia,Brazil,,,UAE,Malaysia,Philippines,,,,
704	630720	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
705	630790	Australia,Korea,Russia,UAE,Malaysi a,Singapore,Thailand,South Africa	
706	631090		Thailand,,
707	640110		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, Singapore, The State St

			outh Africa,
708	640312		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
709	640320	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
710	640351	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa.
711	640391	Korea,UAE	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
712	640420	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South A frica
713	640610	Singapore, Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
714	650100		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
715	670300		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
716	680221	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
717	680223	UAE,Singapore,South Africa	Australia, Brazil, Korea, Russia, Malaysia, Philippines, , Thailand, ,
718	680229		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
719	680299	UAE,Thailand	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, South Africa,
720	680300		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa.
721	680410		Australia,Brazil,Korea,,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa.
722	680423		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
723	681130		Australia,Brazil,Korea,Russia,,,,,,
724	681250		,Brazil,,Russia,,,,,,
725	681260		Australia,Brazil,Korea,Russia,,,,Singapore,,,
726	681270		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
727	681290	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
728	681410	Korea	Australia,Brazil,,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa,
729	681490		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
730	690320	Korea	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
731	690710	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
732	690810		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
733	691090	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
734	700420		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
735	701010		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
736	701190		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
737	701590		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
738	701610		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
739	701790		Thailand,,
740	701810	Australia,Korea	,Brazil,,Russia,,Malaysia,Philippines,,Thailand,South Africa,
741	701990	Korea,UAE,Malaysia,Thailand	Australia,Brazil,,Russia,,,Philippines,,,South Africa,
742	710231	Thailand	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,,
743	710239	Australia,Korea,Singapore,Thailand,S outh Africa	,Brazil,,Russia,,Malaysia,Philippines,,,,
744	710310	Australia	,Brazil,,Russia,,Malaysia,Philippines,,,,
745	710391	Brazil,Singapore,Thailand	Australia,,Korea,Russia,,Malaysia,Philippines,,,,

746	710399	Australia,Brazil,South Africa	Korea, Russia, Malaysia, Philippines, ,,,
747	710510		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
748	711311	Australia,UAE,Singapore,Thailand	Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
749	711319	Australia,Korea,UAE,Singapore,Sout h Africa	Brazil,,Russia,,Malaysia,Philippines,,,,
750	711411	Singapore, Thailand	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
751	711419		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
752	711711	UAE,Malaysia	Australia, Brazil, Korea, Russia, ,, Philippines, Singapore, Thailand, South Africa,
753	711719	UAE,Malaysia,Singapore,Thailand	Australia,,,Russia,,,Philippines,,,South Africa,
754	711790	Australia,Korea,Malaysia,Singapore	Russia,,South Africa,
755	720110		Australia,Brazil,Korea,Russia,UAE,,,Singapore,,South Africa,
756	720120		Australia,Brazil,Korea,Russia,,,,,,
757	720150		Australia,Brazil,Korea,Russia,,,,,,
758	720211		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
759	720230	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
760	720241		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
761	720249		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
762	720310		Australia,Brazil,Korea,Russia,,,,,,
763	720390		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
764	720529		Thailand,,
765	720610		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
766	720690		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
767	720719	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
768	720810		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
769	720825	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
770	720837	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
771	720839		Australia,Brazil,,Russia,,Malaysia,Philippines,,,South Africa,
772	720854		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
773	720915	Singapore	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines,, Thailand,,
774	720928		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
775	720990	Philippines	Australia, Brazil, Korea, Russia, UAE, Malaysia, , Singapore, Thailand, South Africa,
776	721011	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
777	721041	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
778	721049	UAE,Malaysia,Philippines	Australia,Brazil,,Russia,,,,,South Africa,
779	721061	UAE	Australia,,,,,Malaysia,Philippines,,,South Africa,
780	721090	Australia,South Africa	Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,,
781	721129	Thailand	Australia, Brazil, Korea, ,UAE, Malaysia, Philippines, Singapore, ,South Africa,
782	721610		Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,,
783	721669		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
784	721899		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
785	721911		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
786	721914	Korea	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
787	722011		Thailand

788	722090	UAE	Brazil,,,South Africa,
789	722100	UAE,Malaysia,Philippines	Australia,Brazil,Korea,Russia,,,,,South Africa,
790	722220	Korea,Russia,Singapore	Australia,Brazil,,,UAE,Malaysia,Philippines,,Thailand,South Africa,
791	722240	UAE,South Africa	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand,
792	722300	Australia,Korea,Russia,UAE,Singapo	,Brazil,,,,Malaysia,Philippines,,Thailand,South Africa,
793	722620		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
794	722810	Singapore	,Brazil,Korea,Russia,,,,,South Africa,
795	722860		,Brazil,Korea,Russia,,,,Singapore,,,
796	722880		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
797	730120	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
798	730300	Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
799	730410	Korea	Australia,,,Russia,,Malaysia,Philippines,,Thailand,South Africa,
800	730421	Singapore	,Brazil,Korea,Russia,,,,,South Africa,
801	730511	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
802	730519		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,South Africa.
803	730590	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa
804	730610	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa
805	730690	Korea,Malaysia	Australia,Brazil,,Russia,,,Philippine
806	730711	Australia,Korea,UAE	,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
807	730721	Korea,UAE,Singapore	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa
808	730729	Thailand	
809	730791	Australia,UAE,Singapore	,Brazil,,,,Malaysia,Philippines,,Thailand,South Africa
810	730820		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, Singapore, Thailand, S
			outh Africa,
811	731100	UAE,Malaysia,Singapore,Thailand,S outh Africa	Australia,Philippines
812	731512	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
813	731811	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
814	731990	UAE	,Brazil,Korea,Russia,,,,,,South Africa,
815	732010	Korea, Philippines, Singapore, Thailand	Australia,,,Russia,,Malaysia,,,,South Africa,
816	732391	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
817	732393	Korea, UAE, Malaysia, Philippines, Sin	Australia,Brazil,,Russia,,,,,,
818	732394	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
819	732591		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
820	732599	Korea,UAE,Malaysia,Singapore,Thail and	Australia,,,Russia,,,Philippines,,,South Africa,
821	732619	UAE,Malaysia,Philippines,Singapore, Thailand	Australia, South Africa,
822	740110		Australia,Brazil,Korea,Russia,,,,,,
823	740200	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,
824	740311	Singapore	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
825	740321	Australia,UAE	,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
826	740620		Australia,Brazil,Korea,Russia,,,Philippines,,,South Africa,
827	740811	Korea, Malaysia, , Singapore	Australia, Brazil, Russia, ,, Philippines, , Thailand, South Africa,
828	740822		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
829	740921	Philippines	Australia,Brazil,Korea,,,Malaysia,,,,South Africa,
830	741122	Australia	,Brazil,Korea,,,Malaysia,Philippines,Singapore,Thailand,South Africa,
831	741521	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,Thailand,South Africa,

832	741533		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
833	741819	Singapore, Thailand	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
834	741820		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
835	741910		Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
836	741999	Australia,Korea,RussiaUAE,Singapor e,Thailand,South Africa	Brazil,,,,Malaysia,Philippines,,,,
837	750511		Australia,Brazil,,Russia,,Malaysia,Philippines,,,South Africa,
838	750521		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
839	750810		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
840	760110	Malaysia,Singapore	,Brazil,,Russia,,,Philippines,,Thailand,South Africa,
841	760320	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
842	760900	Singapore	Australia, Brazil, Korea, Russia, UAE,, Philippines,, Thailand, South Africa,
843	761210		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
844	761410		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa
845	761490		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa
846	761511		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa
847	780411	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa
848	790111	UAE	Australia, Brazil, ,,, Malaysia, Philippines, Singapore, Thailand, South Africa
849	790112		Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa
850	790120	UAE	Australia, Brazil,,,,,Malaysia, Philippines, Singapore, Thailand, South Africa
851	790310	UAE	Australia,Brazil,Philippines,,,South Africa,
852	790400	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
853	790700	UAE,Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, , Thailand, ,
854	800300		,Brazil,Korea,,UAE,,,,,South Africa,
855	810730		Australia
856	810790		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa
857	811090		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa
858	811251		Australia
859	820120		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
860	820130		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
861	820190	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
862	820310	Russia,UAE	Australia, Brazil, Korea, ,, Malaysia, Philippines, Singapore, Thailand, South Africa,
863	820340	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
864	820411	Australia Korea Russia UAE Singapore Thailand South Africa	Brazil,,,,Malaysia,Philippines
865	820412	Russia UAE Singapore	Australia,Brazil,Korea,,,Malaysia,Philippines,,Thailand,South Africa,
866	820420		Thailand,,
867	820510	Russia UAE	Brazil,Korea,,,,,,South Africa,
868	820570	Australia Russia Singapore Thailand South Africa	Brazil,,,UAE,Malaysia,Philippines,,,,
869	820580	Singapore	,Brazil,Korea,Russia,,,,,,South Africa,
870	820713		Thailand
871	821195		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
872	821210	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,South Africa,
873	821510		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
874	830510		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South
-----	--------	------------------------------	---
875	840212		Africa, Australia Brazil Korea Russia, Malaysia Philippines, South Africa
876	840219	Korea Singapore	Australia Brazil Russia UAE Malaysia Philippines Thailand South Africa
877	840420		Australia Brazil Korea Russia. Malaysia Philippines, Singapore Thailand South
			Africa,
878	840490	Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
879	840590	Korea UAE	Australia, Brazil, Russia, Malaysia, Philippines, Thailand, South Africa,
880	840610		Australia,Brazil,Korea,Russia,,,,,,
881	841320	Australia UAE South Africa	,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,,
882	841440	UAE Singapore	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
883	842099		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
884	842111		Australia,Brazil,Korea,Russia,,,,,,
885	843221		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
886	843590		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
887	843790		Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, ,, South Africa,
888	843830		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
889	844250	Russia UAE Thailand	Australia, Brazil, Korea,,,, Philippines,,, South Africa,
890	844329		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,Thailand,S outh Africa,
891	844610		Thailand,,
892	844629		Australia,Brazil,Korea,Russia,UAE,,,Singapore,,South Africa,
893	844820	Singapore	,Brazil,,Russia,UAE,,,,,South Africa,
894	844831		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
895	844833	UAE	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
896	845230		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
897	845522		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
898	845899	UAE Malaysia	Australia, Brazil, Korea, Russia,,, Philippines,, Thailand, South Africa,
899	846820	Thailand	Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
900	847521		Australia, Brazil, Korea, Russia, Malaysia, Philippines, ,, South Africa,
901	847629		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
902	847751		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
903	847920	Malaysia	Australia,,Korea,Russia,,,Philippines,,,South Africa,
904	847960		Malaysia, Philippines,, Thailand,,
905	848010		Australia, Brazil, Korea, ,UAE, Malaysia, Philippines, Singapore, Thailand, South Africa,
906	848291	Australia Brazil	UAE,Malaysia,Philippines,,Thailand,South Africa,
907	850133	UAE	Australia, Brazil, Korea, Russia, Malaysia, Philippines, Thailand, South Africa,
908	850231		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
909	850421	Brazil Singapore	Australia,,Korea,Russia,UAE,Malaysia,Philippines,,Thailand,South Africa,
910	850422		Australia, Brazil, Korea, "Malaysia, Philippines, Singapore, Thailand, South Africa,
911	850423	Malaysia	Australia, Brazil, Korea, Russia, "Philippines, Singapore, Thailand, South Africa,
912	851230	Korea UAE Malaysia Singapore	Australia,Brazil,,Russia,,,Philippines,,Thailand,South Africa,
913	851420		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
914	853230		,Brazil,Korea,Russia,UAE,,,,,South Africa,

915	853510	Russia	Australia,Brazil,Korea,,UAE,Malaysia,Philippines,,,South Africa,
916	854012		,Thailand,,
917	854419	Malaysia	Australia,Brazil,Korea,Russia,UAE,,Philippines,,,South Africa,
918	854511	Korea South Africa	Australia,Brazil,,Russia,,Malaysia,Philippines,Singapore,Thailand,,
919	854620	Russia Singapore	Australia,Brazil,Korea,,UAE,Malaysia,Philippines,Thailand,South Africa,
920	854690	Australia Brazil UAE Malaysia	Korea, Russia,,,,,,South Africa,
921	854790	Australia Brazil Russia UAE	Korea,,,,Malaysia,Philippines,,Thailand,South Africa,
922	860110	Singapore	Australia
923	860120		Australia,
924	870130		Thailand,,
925	870321	UAE	Australia,,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South Africa,
926	870600	Malaysia	Australia,Brazil,Korea,Russia,UAE,,Philippines,,,South Africa,
927	870810	Australia Brazil Korea UAE Malaysia South Africa	",Russia,,,Philippines,,,,
928	870990		,Brazil,Korea,Russia,UAE,,,,,South Africa,
929	871492	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
930	871493		Arrica, Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines, South Africa.
931	890110	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
932	890130		Australia,,,,,,,
933	890400		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,,,,
934	890510		Australia,Brazil,Korea,Russia,,,,Singapore,,South Africa,
935	890520		Brazil,Korea,,UAE,,,,,South Africa,
936	890590		Australia,Brazil,Korea,,UAE,Malaysia,Philippines,,South Africa,
937	890610		Australia,,,,,,,
938	890690		Australia,Brazil,Korea,Russia,UAE,Malaysia,Philippines,,,South Africa,
939	900140	Philippines Singapore	Australia,Brazil,Korea,Russia,UAE,Malaysia,,,,South Africa,
940	901820		,,,,,,Thailand,,
941	902221		Australia,Brazil,Korea,Russia,UAE,,,Singapore,,South Africa,
942	902230	Singapore	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Thailand,South Africa,
943	910191		Thailand
944	910219	UAE	Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
945	910529	UAE	Atrica, Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,Thailand,South
0.16	011010		Africa,
946	911019		Australia
947	911410		Australia,Brazil,Korea,Russia,,Malaysia,,Singapore,Thailand,South Africa,
948	911430		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
949	920992		,,,,,,Thailand,,
950	930111		,,,,,,Thailand,,
951	940410		Australia,Brazil,Korea,Russia,,Malaysia,Philippines,Singapore,,South Africa,
952	960190	Korea	Australia,Brazil,,Russia,,Malaysia,,Singapore,Thailand,South Africa,
953	960200	Thailand	Australia,Brazil,,Russia,UAE,Malaysia,Philippines,Singapore,,South Africa,
954	960630	Korea	Australia, Brazil, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
955	960831		Australia, Brazil, Korea, Russia, Malaysia, Philippines, Singapore, Thailand, South Africa,
956	960891	Korea Malaysia	Australia,Brazil,,Russia,,,Philippines,Singapore,,South Africa,
957	970190	Singapore South Africa	Australia, Brazil, Korea, Russia, UAE, Malaysia, Philippines,, Thailand,,
958	970300	Thailand	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,